

\documentclass[12pt]{amsart}

\newfont{\cyr}{wncyr10 scaled 1100}

\usepackage[left=2.7cm,right=2.7cm,top=3.5cm,bottom=3cm]{geometry}
\usepackage{amsthm,amssymb,amsmath,amsfonts,mathrsfs,amscd, graphics, color}
\usepackage[latin1]{inputenc}
\usepackage[all]{xy}
\usepackage{latexsym}
\usepackage{longtable}
\usepackage{color}
\usepackage{tikz}
\usepackage{hyperref}



\setcounter{tocdepth}{2}


 
\theoremstyle{plain}
\newtheorem{theorem}{Theorem}[section]
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{proposition}[theorem]{Proposition}
\newtheorem{conjecture}[theorem]{Conjecture}

\theoremstyle{definition}
\newtheorem{definition}[theorem]{Definition}
\newtheorem{question}[theorem]{Question}
\newtheorem{examplewr}[theorem]{Example}
\newtheorem{ass}[theorem]{Assumption}
\newtheorem{app}[theorem]{Application}

\theoremstyle{remark}
\newtheorem{obswr}[theorem]{Observation}
\newtheorem{remarkwr}[theorem]{Remark}
\newtheorem{example}[theorem]{Example}



\newenvironment{obs}{\begin{obswr}\begin{upshape}}{\end{upshape}\end{obswr}}
\newenvironment{remark}{\begin{remarkwr}\begin{upshape}}{\end{upshape}\end{remarkwr}}
% \newenvironment{example}{\begin{examplewr}\begin{upshape}}{\end{upshape}\end{examplewr}}

\DeclareMathOperator{\fil}{fil}
\DeclareMathOperator{\fr}{fr}
 \DeclareMathOperator{\w}{w}




\DeclareMathOperator{\an}{an}
\DeclareMathOperator{\nh}{nh}
\DeclareMathOperator{\cusps}{cusps}
\DeclareMathOperator{\ffm}{ffm}
\DeclareMathOperator{\ppar}{par}
\DeclareMathOperator{\alg}{alg}
\DeclareMathOperator{\Ext}{Ext}



\DeclareMathOperator{\lcm}{lcm}
\DeclareMathOperator{\dR}{dR}
\DeclareMathOperator{\fp}{\mathfrak{p}}
\DeclareMathOperator{\cris}{cris}
\DeclareMathOperator{\Frac}{Frac}
\DeclareMathOperator{\et}{et}
\DeclareMathOperator{\syn}{syn}
\DeclareMathOperator{\oc}{oc}
\DeclareMathOperator{\ur}{u-r}
\DeclareMathOperator{\ah}{a-h}
\DeclareMathOperator{\para}{par}
\newcommand{\hV}{{\mathbb{V}}}
\newcommand{\hU}{{\mathbb{U}}}
\newcommand{\hW}{{\mathbb{W}}}
\DeclareMathOperator{\nr}{nr}
\newcommand{\mf}{{\mathfrak{f}}}


\DeclareMathOperator{\cts}{cts}
\DeclareMathOperator{\bal}{bal}
\DeclareMathOperator{\KS}{KS}
\DeclareMathOperator{\sym}{\mathrm{sym}}
\DeclareMathOperator{\Sym}{Sym}
\DeclareMathOperator{\Ad}{Ad}
\DeclareMathOperator{\sgn}{sgn}
\DeclareMathOperator{\spec}{spec}
\DeclareMathOperator{\la}{la} 
\DeclareMathOperator{\Ind}{Ind}
\DeclareMathOperator{\ad}{Ad}

\DeclareMathOperator{\shi}{{S-H_1}}
\DeclareMathOperator{\shii}{S-H_2}

\def\rhom{{V_{_{\!\heartsuit}}}}


\newcommand{\GoneN}{\Gamma_1(N)}
\newcommand{\XoneN}{X_1(N)}
\newcommand{\Mk}{{\mathcal M}_k(\Gamma_1(N))}
\newcommand{\Sk}{{\mathcal S}_k(\Gamma_1(N))}

\newcommand{\fD}{{\mathfrak{D}}}
\newcommand{\fG}{{\mathfrak{G}}}

\newcommand{\fM}{{\mathfrak{M}}}


\newcommand{\Res}{{\rm Res}}

\newcommand{\hiddensubsection}[1]{
    \stepcounter{subsection}
    \subsection*{\arabic{section}.\arabic{subsection}\hspace{1em}{#1}}
}



\newcommand{\uomega}{{\underline{\omega}}}
\newcommand{\cZ}{{\mathcal Z}}
\newcommand{\hf}{{\mathbf{f}}}
\newcommand{\hg}{{\mathbf{g}}}
\newcommand{\hh}{{\mathbf {h}}}
\newcommand{\hE}{{\mathbf {E}}}
% For below, try \breve, \check, \dot, \hat, \mathring, \ddot
\newcommand{\testF}{\breve{F}}
\newcommand{\testphi}{\breve{\phi}}
\newcommand{\testf}{\breve{f}}
\newcommand{\testg}{\breve{g}} 
\newcommand{\testh}{\breve{h}}
\newcommand{\testgamma}{\breve{\gamma}}
\newcommand{\testG}{\breve{G}}
\newcommand{\testH}{\breve{H}}
\newcommand{\htf}{\breve{\mathbf{f}}}
\newcommand{\htg}{\breve{\mathbf{g}}}
\newcommand{\hth}{\breve{\mathbf {h}}}
\newcommand{\uS}{{\mathbf{S}}}
\newcommand{\hphi}{\boldsymbol{\phi}}
\newcommand{\uhp}{{\mathfrak{H}}}
\newcommand{\PPhi}{{\underline{\phi}}}
\newcommand{\ff}{{\mathbf{f}}}
\newcommand{\aaa}{{\mathbf{a}}}
\newcommand{\LL}{{\mathscr{L}}}
\newcommand{\Lp}{{\mathscr{L}_p}}
\newcommand{\rhop}{\rho(P,\omega_2,\omega_3)}
\newcommand{\ukappa}{\underline{\kappa}}
\newcommand{\uV}{{\bf V}}

 
\newcommand{\fa}{{\mathfrak{a}}}
\newcommand{\fb}{{\mathfrak{b}}}
\newcommand{\fc}{{\mathfrak{c}}}
\newcommand{\fd}{{\mathfrak{d}}}

\newcommand{\Ver}{{\mathrm{Ver}}}

\newcommand{\dd}{{\underline{d}}}

\newcommand{\hB}{{\mathbf B}}
\newcommand{\St}{{(\infty)}}
\newcommand{\cl}{{\mathrm{cl}}}
\DeclareMathOperator{\red}{red}

\newcommand{\Nfgh}{{N_{_{\!fgh}}}}

\newcommand{\cA}{\mathcal A}
\newcommand{\cN}{\mathcal N}
\newcommand{\cW}{\mathcal W}
\newcommand{\cV}{\mathcal V}
\newcommand{\cF}{\mathcal F}
\newcommand{\cX}{\mathcal X}
\newcommand{\cH}{\mathcal H}
\newcommand{\cC}{\mathcal C}
\newcommand{\V}{\mathcal V}
\newcommand{\E}{\mathcal E}
\newcommand{\cE}{\mathcal E}
\newcommand{\g}{\gamma}
% \newcommand{\G}{\Gamma}
\newcommand{\Gap}{\Gamma_0^D(p M)}
\newcommand{\Ga}{\Gamma_0^D(M)}
\newcommand{\hGa}{\hat{\Gamma }_0(M)}
\newcommand{\Q}{\mathbb{Q}}
\newcommand{\A}{\mathbb{A}}
\newcommand{\Z}{\mathbb{Z}}
\newcommand{\G}{\mathbb{G}}
\newcommand{\F}{\mathbb{F}}
\newcommand{\K}{\mathbb{K}}
\newcommand{\C}{\mathbb{C}}
\newcommand{\I}{\mathbb{I}}
\newcommand{\Ql}{\Q_{\ell}}
\newcommand{\GQ}{G_{\Q}}
\newcommand{\PP}{\mathbb{P}}
\newcommand{\Sel}{\mathrm{Sel}}
\newcommand{\Sha}{\mathrm{Sha}}
\newcommand{\Gal}{\mathrm{Gal\,}}
\newcommand{\GL}{\mathrm{GL}}
\newcommand{\Cl}{\mathrm{Cl}}
\newcommand{\Tr}{\mathrm{Tr}}
\newcommand{\Div}{\mathrm{Div}}
\newcommand{\Fil}{\mathrm{Fil}}
\newcommand{\cond}{\mathrm{cond}}
\newcommand{\Frob}{\mathrm{Fr}}
\newcommand{\End}{\mathrm{End}}
\newcommand{\Jac}{\mathrm{Jac}}
\newcommand{\disc}{\mathrm{disc}}
\newcommand{\Id}{\mathrm{Id}}
\newcommand{\MS}{\mathrm{MS}}
\newcommand{\Aut}{\mathrm{Aut}}
\newcommand{\Fr}{\mathrm{Fr}}
\newcommand{\diag}{\mathrm{diag}}
\newcommand{\Up}{\mathrm{U}_p}
\newcommand{\ord}{{\mathrm{ord}}}
\newfont{\gotip}{eufb10 at 12pt}
\newcommand{\gn}{\mbox{\gotip n}}
\newcommand{\gm}{\mbox{\gotip m}}
\newcommand{\ga}{\mbox{\gotip a}}
\newcommand{\gb}{\mbox{\gotip b}}
\newcommand{\gf}{\mbox{\gotip f}}
\newcommand{\go}{{\mathcal O}_{A_f}}
\newcommand{\OAf}{{\mathcal O}_{A_f}}
\newcommand{\gp}{\mbox{\gotip p}}
\newcommand{\gL}{\mbox{\gotip L}}
\newcommand{\gd}{\mbox{\gotip d}}
\newcommand{\cO}{{\mathcal O}}
\newcommand{\cM}{{\mathcal M}}
\newcommand{\cP}{{\mathcal P}}
\newcommand{\cL}{{\mathcal L}}
\newcommand{\cG}{{\mathcal G}}
\newcommand{\GG}{{\mathcal G}}
\newcommand{\cD}{{\mathcal D}}
\newcommand{\ra}{\rightarrow}
\newcommand{\lra}{\longrightarrow}
\newcommand{\hra}{\hookrightarrow}
\newcommand{\qbar}{\bar\Q}
\newcommand{\Norm}{\mathrm{N}}
\newcommand{\NS}{\mathrm{NS}}
\newcommand{\SL}{{\mathrm {SL}}}
\newcommand{\SO}{{\mathrm {SO}}}
\newcommand{\n}{{\mathrm{n}}}
\newcommand{\Pic}{{\mathrm{Pic}}}
\newcommand{\CH}{{\mathrm{CH}}}
\newcommand{\AJ}{{\mathrm{AJ}}}
\newcommand{\tr}{{\mathrm{tr}}}
\newcommand{\si}{\sigma}
\newcommand{\eps}{\epsilon}
\newcommand{\ta}{{\tau}}
\newcommand{\R}{{\mathbb R}}
\newcommand{\M}{{\mathrm{M}}}
\newcommand{\HC}{\mathcal F_{\rm har}}
\newcommand{\PGL}{{\mathrm{PGL}}}
\newcommand{\Supp}{\mathrm{Supp}}
\newcommand{\ep}{\varepsilon}

\def \mint {\times \hskip -1.1em \int}

%\newcommand{\mint}{\times\!\!\!\!\!\hskip-0.3em\int}

\newcommand{\SP}{\vspace{0.5cm}}
\newcommand{\T}{\mathbb T}
\newcommand{\new}{\mathrm{new}}
\DeclareMathOperator{\Hom}{Hom} 
\DeclareMathOperator{\Meas}{Meas}
\DeclareMathOperator{\cyc}{cyc}
\newcommand{\X}{\mathbb X}
\newcommand{\Y}{\mathbb Y}
\newcommand{\U}{\mathbb U}
\newcommand{\Rat}{{\rm Rat}\bigl(\PP^1(\Q_p),\C_p\bigr)}
\newcommand{\Ker}{{\mathrm Ker}}
\newcommand{\res}{\mathrm{res}}
\DeclareMathOperator{\rig}{rig}
\DeclareMathOperator{\cmp}{comp}
\DeclareMathOperator{\can}{can}
\DeclareMathOperator{\fin}{fin}
\DeclareMathOperator{\even}{even}
\DeclareMathOperator{\spl}{Spl}
\DeclareMathOperator{\hdg}{hdg}
 \DeclareMathOperator{\noc}{n-oc}
 \DeclareMathOperator{\hol}{hol}
 \newcommand{\Corr}{\mathrm{Corr}}
 \DeclareMathOperator{\As}{As}
 

\newcommand{\TT}{\mathbb{T}}
\newcommand{\uTT}{\underline{\mathbb{T}}}
\newcommand{\D}{\mathbb D}


\newcommand{\rom}{\mathrm}
 

\newcommand{\longmono}{\mbox{$\lhook\joinrel\longrightarrow$}}
\newcommand{\longleftmono}{\mbox{$\longleftarrow\joinrel\rhook$}}
\newcommand{\longepi}{\mbox{$\relbar\joinrel\twoheadrightarrow$}}
\newcommand{\mat}[4]{\left(\begin{array}{cc}#1&#2\\#3&#4\end{array}\right)}
\newcommand{\smallmat}[4]{\bigl(\begin{smallmatrix}#1&#2\\#3&#4\end{smallmatrix}\bigr)}
\newcommand{\dirlim}{\mathop{\varinjlim}\limits}
\newcommand{\invlim}{\mathop{\varprojlim}\limits}

\def\thetak{\theta^{^{(2k)}}}
\def\thetam{\theta^{^{(2m)}}}
\def\thetakk{\theta^{^{(2k+2)}}}

\def\thetakv{\theta^{^{(2k)\vee}}}
\def\thetamv{\theta^{^{(2m)\vee}}}
\def\thetakkv{\theta^{^{(2k+2)\vee}}}

\def\Bellaiche{ Bella{\"\i}che }

\def\bN{{\mathbf N}}
\def\Gammatau{{\Gamma_{\!\tau}}}
\def\tkappa{{\tilde\kappa}}
\def\ukappa{{\underline\kappa}}
\def\utheta{{\mbox{\boldmath$\theta$}}}
\def\trho{{\tilde\rho}}
\def\lra{{\longrightarrow}}
\def\SL{{\bf SL}}
%\def\GL{{\bf GL}}
\def\PGL{{\bf PGL}}
\def\PSL{{\bf PSL}}
\def\PP{{\mathbf P}}
\def\X{\mathbf{X}}
\def\rp1{r\!\!+\!\!1}
\def\Lp{{\mathscr L}_p}
\newcommand{\om}{{\omega}}
\def\bp{{x}}
\def\bpp{{y}}

% \def\psim{{\psi_{_{\!\!-}}}}
\def\psim{{\psi_{_{\!\heartsuit}}}}
\def\chiK{{\chi_{_{K}}}}

\def\chipp{\chi_{gh}^{\scalebox{0.6}{++}}}
\def\chipm{\chi_{gh}^{\scalebox{0.7}{+}\scalebox{0.9}{-}}}
\def\chimp{\chi_{gh}^{\scalebox{0.9}{-}\scalebox{0.7}{+}}}
\def\chimm{\chi_{gh}^{\scalebox{1}{-}\scalebox{1}{-}}}

 
\def\gauss{\mathfrak{g}}

%% ALAN'S COMMANDS

\newcommand{\lambdatheta}{\lambda_{\theta}}
\newcommand{\lambdahida}{\lambda_{{\rm Hida}}}
%% \newcommand{\lambdanonhida}{\lambda_{{\rm non-Hida}}}
\newcommand{\lambdanonhida}{\lambda_{{\rm inert}}}

 \newcommand{\logtate}{\log_{{\rm Tate}}}
%% \newcommand{\logtate}{\log_{E,p}}



\begin{document}
\begin{thebibliography}{FLPSW}

\bibitem[ABSV]{ABSV}
Fabrizio Andreatta, Massimo Bertolini, Marco Seveso, and Rodolfo Venerucci, {\em Endoscopy for {\bf GSp}$(4)$ and
rational points of elliptic curves}, in preparation

\bibitem[BF20]{BF} Ivan Blanco-Chac\'on and Michele Fornea, 
{\em Twisted triple product $p$-adic $L$-functions and Hirzebruch-Zagier cycles}, Journal of the Institute of Mathematics of Jussieu, 
{\bf 19} (6) (2020) 1947--1992.

\bibitem[BD05]{BD1}
Massimo Bertolini and
Henri Darmon. 
{\em Iwasawa's Main Conjecture for elliptic curves over anticyclotomic $\Z_p$-extensions}, Annals of Math.
{\bf 162} (2005) 1--64.	
 

\bibitem[BD07]{BD2}
Massimo Bertolini and Henri Darmon, {\em Hida families and rational points on elliptic curves}, Inventiones Math. {\bf 168} (2007), no.2, 371-431.

\bibitem[BD09]{BD3}
Massimo Bertolini and  Henri Darmon, {\em The rationality of Stark-Heegner points over genus fields of real quadratic fields}, Annals of Math. {\bf 170} (2009), 343-369.

\bibitem[BL17]{BL}
Kevin Buzzard and Alan Lauder, {\em A computation of modular forms of weight one and small level}, Annales Math\'ematiques du Qu\'ebec {\bf 41}, Issue 2, (2017), 213-219.  Accompanying webpage: {\rm https://people.maths.ox.ac.uk/lauder/weight1/}



\bibitem[CC19]{CC} 
S. Chen, Y. Cheng, {\em On Deligne's conjecture for certain
automorphic L-functions for $\GL(3) \times \GL(2)$ and $\GL(4)$}, Documenta Mathematic {\bf 24} (2019), 2241-2297.

\bibitem[DaH]{dallava-horawa}
Luca Dall'Ava and Alex Horawa. {\em 
Balanced triple product $p$-adic $L$-functions and Stark points},
submitted.



\bibitem[Dar00]{Dar} Henri Darmon, {\em Integration on $\cH_p\times\cH$ and arithmetic applications}, Annals of Math. {\bf 154}  (2000), 589-639.

\bibitem[DLR15]{DLR1}
Henri Darmon, Alan Lauder, and Victor Rotger,  
{\em Stark points and $p$-adic iterated integrals attached to modular forms of weight one.}
Forum of Mathematics, Pi, (2015), Vol.~3, e8, 95 pages.

\bibitem[DPV21]{DPV}
Henri Darmon, Alice Pozzi, and Jan Vonk, {\em Diagonal restrictions of $p$-adic Eisenstein families}, Math.\,Annalen {\bf 379} (2021), 503--548.  


\bibitem[DR14]{DR1}
Henri Darmon and  Victor Rotger, 
{\em Diagonal cycles and Euler 
systems I: a $p$-adic Gross-Zagier formula}, 
 Annales Scientifiques 
de l'Ecole Normale Sup\'erieure {\bf 47}, n. 4 (2014), 779--832.


\bibitem[DR17]{DR2}
Henri Darmon and Victor Rotger, {\em 
Diagonal cycles and Euler systems II:   the Birch and Swinnerton-Dyer conjecture
 for Hasse-Weil-Artin $L$-series},
Journal of the American Mathematical Society {\bf 30} Vol. 3, (2017), 601-672. 


\bibitem[FLP+24]{pozzi-win}
Paulina Fust, Judith Ludwig, 
Alice Pozzi, Mafalda Santos and Hanneke Wiersema,  {\em Real quadratic singular moduli and $p$-adic families of modular forms},    Women in Numbers Europe IV, Association for Women in Mathematics Series {\bf 32}, (2024),  253-291.

\bibitem[GZ86]{gross-zagier}
Benedict Gross and  Don Zagier, {\em
Heegner points and derivatives of $L$-series}. 
Invent. Math. {\bf 84} (1986), no. 2, 225--320.


 \bibitem[HM25]{HM}
Victor Hernandez and Santiago Molina, {\em Plectic points and Hida-Rankin p-adic L-functions}, Transactions of the American Mathematical Society {\bf 387}, no. 9, (2025), 6535-6583.

\bibitem[Hi94]{Hida}
Haruzo Hida, {\em On the critical values of $L$-functions of $\GL(2)$ and $\GL(2)\times \GL(2)$}, Duke Math.\,J. {\bf 74} (1994), 431--529.

\bibitem[Hs21]{Hs} Ming-Lun Hsieh, {\em Hida families and $p$-adic triple product $L$-functions}, American Journal of Mathematics {\bf 143}, no.2, (2021), 411-532.



\bibitem[Ich08]{Ich} Atsushi Ichino, {\em Trilinear forms and the central values of triple product L-functions}, Duke
Math. J., {\bf 145 (2)} (2008), 281--307.


\bibitem[Ish25]{Ish} I.\,Ishikawa, {\em On the construction of twisted triple product $p$-adic $L$-functions for unbalanced weights}, preprint 2025,
available at \url{https://researchmap.jp/1sa014kawa/misc/53586885?lang=en}. 

\bibitem[Kat04]{Ka}
Kazuya Kato, {\em $p$-adic Hodge 
theory and values of zeta functions of modular forms},  Ast\'erisque {\bf 295}, 2004.

\bibitem[Kaz24]{Kazi}
Ananyo Kazi, {\em Twisted triple product $p$-adic $L$-function for finite slope families of Hilbert modular forms}, preprint 2024.


\bibitem[KL25]{KL}
Ananyo Kazi and David Loeffler, {\em $p$-adic Asai and twisted triple product $L$-functions for finite slope families}, preprint 2025.


\bibitem[Ko88]{kolyvagin}
Victor Kolyvagin, 
{\em Finiteness of $E(\Q)$ 
and $L\!\!L\!\!I(E/\Q)$  for a subclass of Weil curves.}
Izv. Akad. Nauk SSSR Ser. Mat. {\bf 52} (1988), no. 3, 522--540, 670--671.
Math. USSR-Izv. {\bf 32} (1989), no. 3, 523--541.

\bibitem[La11]{AL1}
Alan Lauder, Computations with classical and $p$-adic modular forms, LMS J. Comput. Math. {\bf 14} (2011), 214-231

\bibitem[La14]{AL2}
Alan Lauder, Efficient computation of Rankin $p$-adic L-functions, in ``Computations with Modular Forms. Proceedings of
a Summer School and Conference, Heidelberg, August/September 2011'', Boeckle G. and Wiese G. (eds), Springer Verlag,
181-200, 2014.

\bibitem[LMFDB]{LMFDB} The LMFDB Collaboration, The L-functions and modular forms database, {\rm https://www.lmfdb.org}



\bibitem[LV14]{LV}
Matteo Longo and Stefano Vigni,  {\em The rationality of quaternionic Darmon points over genus fields
 of real quadratic fields}. Int. Math. Res. Not. {\bf 13} (2014), 3632--3691. 

 
\bibitem[Mok11]{Mok} Chung-Pang Mok, {\em Heegner points and p-adic L-functions for elliptic curves over
certain totally real fields}. Comment. Math. Helv., {\bf 86} (4), (2011), 867--945.

\bibitem[Pra92]{prasad}
Dipendra Prasad, {\em Invariant forms for representations of $\GL_2$ over a local field}, Amer. J. Math.
{\bf 114} (1992), no. 6, 1317--1363.


\end{thebibliography}


\include{thebibliography}
 

 
%% \title[An elliptic Stark conjecture
%% for Asai representations]{An elliptic Stark conjecture \\
%%for Asai representations}
\title[Stark-Heegner points for Asai representations] {Stark-Heegner points for Asai representations}



\author{Henri Darmon, Alan Lauder and Victor Rotger}


\begin{abstract} 
A conjectural construction of global points on modular abelian varieties is proposed. These points  are defined over the field  cut out by the  tensor induction
(or Asai representation) of a totally  odd two-dimensional Galois representation  of a real quadratic field.
     \end{abstract}
 
 
 
 

\address{H. D.: McGill University, Montreal, Canada}
\email{darmon@math.mcgill.ca}
\address{A. L.: University of Oxford, U. K.}
\email{alan.lauder@maths.ox.ac.uk}
\address{V. R.: Universitat Polit\`{e}cnica de Catalunya, Barcelona, Spain}
\email{victor.rotger@upc.edu}

%\subjclass{11G18, 14G35}


\maketitle

\tableofcontents

 
\section*{Introduction} 

Let $E$ be an elliptic curve over $\Q$ and let $V$ be an Artin representation of $\Q$, that is to say, a finite-dimensional vector space over  a finite extension $\kappa\subset \C$ of $\Q$, called the {\em field of coefficients} of $V$, endowed with a continuous linear action of $G_{\Q}:= {\rm Gal}(\bar\Q/\Q)$. The action of $G_\Q$ factors through the quotient ${\rm Gal}(H/\Q)$ for a finite extension $H$ of $\Q$, referred to as the field {\em cut out} by $V$, giving rise to a homomorphism
 $$
 \varrho:  {\rm Gal}(H/\Q) \,\hookrightarrow \, \GL(V).
 $$
 For each prime $\ell$, denote by $\kappa_\ell$   a completion of $\kappa$ at a prime above $\ell$. 
The pair $(E,V)$ gives rise to a
 continuous $\kappa_\ell$-linear representation of $G_\Q$
 whose underlying vector space   is
 the tensor product  of the Tate module of $E$ with $V$:
\begin{equation}
 W_{E,V,\ell}:= T_\ell(E) \otimes_{ \kappa_\ell}  (V\otimes_\kappa \kappa_\ell), \quad \mbox{ where } \quad T_\ell(E) := (\lim_{n, \leftarrow} E[\ell^n])\otimes_{\Z_\ell} \kappa_\ell.
 \end{equation}
 It is unramified at any rational prime
 $p$   not dividing $ \ell N D$, where $N$ and $D$ are the conductors of $E$ and $V$ respectively. The frobenius element at such a  $p$, denoted $\sigma_p$, is well-defined up to conjugation in ${\rm Aut}(W_{E,V,\ell})$. Its characteristic polynomial 
 has coefficients in $\kappa\subset \kappa_\ell$ and does not depend on the choice of $\ell\ne p$.
 
 
 The {\em equivariant Birch and Swinnerton-Dyer conjecture} relates the order of vanishing at $s=1$ of the Hasse-Weil-Artin $L$-series
 \begin{equation}
 L(E,V,s) := \prod_{p\nmid ND}\det(1-\sigma_p \cdot p^{-s})^{-1} 
 \end{equation}
to the $V$-rank of $E$, defined as
\begin{equation}\label{EV}
 {\rm rank}(E,V) := 
\dim_{\kappa} E(H)^V, \quad \mbox{ where } \quad E(H)^V:= \hom_{G_\Q}(V, E(H)\otimes \kappa).
\end{equation}

\medskip \medskip
\noindent
{\bf Conjecture BSD(E,V)}:   
$ \ord_{s=1} L(E,V,s) = {\rm rank}(E,V).$  

\medskip \medskip
\noindent
Progress on this conjecture has  been painstaking and hard-won.
In the analytic rank $0$ situation where  $L(E,V,1)\ne 0$,  it is known when 
$V$ is one-dimensional \cite{Ka},  when $V$ is induced from certain ring class characters of an imaginary quadratic  field \cite{BD1}
or   a real quadratic field \cite{DR2}, or when 
 $V$ is the tensor product of two odd irreducible two-dimensional Artin representations \cite{DR2}.
 
 \medskip
 Results in the analytic rank one case, where $L(E,V,s)$ has a simple zero at $s=1$, are even more fragmentary  and  largely confined to scenarios where the vanishing of $L(E,V,s)$  is forced by  parity considerations. This happens  when $V$ is isomorphic to its contragredient--- so that the functional equation relates the values of $L(E,V,s)$  at $s$ and $2-s$---and when the sign in this functional equation, denoted ${\rm sign}(E,V)$, is equal to $-1$.
The $L$-function $L(E,V,s)$ is 
  then essentially an odd function of $s-1$, and hence vanishes to odd order at the central point. 
  
 Assuming for simplicity that 
  $N$ and $D$ are relatively
  prime to each other, the most
  basic instances of this  occur when the representation
  $V$ is induced from a ring class character of a  quadratic field $F$. In that case, $V$ is always self-dual and  
  \begin{equation}
  \label{eqn:sign-rcc}
  {\rm sign}(E,V) = \varepsilon_F(-N),
  \end{equation}
  where $\varepsilon_F$ is the quadratic Dirichlet character attached to $F$.
   \begin{itemize}
  \item[(a)]  When $F$ is imaginary and $\varepsilon_F(-N)=-1$,
  the  systematic vanishing of $L(E,V,1)$   is accounted for by a plentiful supply of complex multiplication points defined over all ring class fields of $F$  of conductor prime to $N$, lying  on suitable 
 modular or Shimura curves which uniformise $E$.
 The significance of the resulting  {\em Heegner points} in $E(F^{\rm ab})$ for the study of the  Birch and Swinnerton Dyer conjecture can hardly be overstated. These points are the basis for the important results of Gross-Zagier \cite{gross-zagier}
 and Kolyvagin \cite{kolyvagin}, and provide essentially  the only instance where a simple zero of $L(E,V,s)$ at $s=1$ 
 can be parlayed into the construction of a non-trivial element 
 of $E(H)^V$.  
 \item[(b)] When $F$ is   real  and $\varepsilon_F(N)=-1$, the role of Heegner points  
is ostensibly played by the {\em Stark-Heegner points} introduced in \cite{Dar}, whose conjectural nature prevents their being used to prove BSD($E,V$) in this case. 
\end{itemize}

 The article \cite{DR2} does prove some non-trivial instances of BSD($E,V$) in  the analytic rank zero variant  of scenario (b),  where $\varepsilon_F(N)=1$. The proof in this case
  is somewhat roundabout and makes no use of Stark-Heegner points: rather it proceeds by realising $V$ as a direct summand of a tensor product $V_1\otimes V_2$
of two odd two-dimensional Artin representations in order to reduce BSD($E,V$) to  BSD($E,V_1\otimes V_2$), which is proved using global cohomology classes in $H^1(\Q, W_{E,V_1\otimes V_2,\ell})$ arising from $p$-adic families of \'etale Abel-Jacobi images of diagonal cycles in triple products of modular curves.   


 A motivation for singling out scenario (b)  for special attention is that it has revealed 
 tantalising  possibilities for constructing global points on elliptic curves in settings that lie squarely beyond the scope of the theory of complex multiplication. It is natural to ask whether other instances of self-dual $V$ for which ${\rm sign}(E,V)=-1$ might lead to analogous---conjectural, but entirely explicit ---constructions
  of global points on $E$.
  
  
 The setting where  $V=V_1\otimes V_2$ is a self-dual tensor product of two odd irreducible two-dimensional Artin representations may seem promising at first glance in light of the progress initiated in \cite{DR2}. However, one  always has
 \begin{equation}
  {\rm sign}(E,V_1\otimes V_2) = 1 \quad \mbox{ when } \quad \gcd(N,D)=1,
  \end{equation}
  which makes this setting  somewhat unpropitious
  for suggesting new Stark-Heegner point constructions.
  (See however \cite{DLR1}
  where  $p$-adic regulators attached to $L(E,V_1\otimes V_2,s)$ are related to appropriate ``Stark points" in favorable ``analytic rank two" scenarios, and the 
   work of Dall'Ava and Horawa \cite{dallava-horawa} which focuses on situations of analytic rank one, where the conductors of $E$ and $V_1\otimes V_2$ are no longer  assumed to be coprime. 
   An ongoing work in progress   of Andreatta, Bertolini, Seveso and Venerucci  \cite{ABSV} approaches  a similar setting   through the study of endoscopic lifts to unitary groups.)
  
  
  
  
  
Motivated by these considerations, this paper explores the  setting where $V$ is the {\em tensor induction}, or  {\em Asai representation},  of a two-dimensional Artin representation   of a real quadratic field $F$. The  definition and key properties of this tensor induction are recalled in
Section \ref{sec:asai-rep}.
 Let
 $ {\rm Ver}_F^\Q$ denote the {\em transfer}
({\em Verlagerung}) of  a character of $G_F$ to a character of $G_\Q$,
which is dual to the transfer map
$ G_\Q^{\rm ab} \lra G_F^{\rm ab}$ 
of   group theory.
  

\bigskip
\noindent
{\bf Sign Formula} (\cite[Thms B \& D, Rk.~4.1.1]{prasad}): {\em Let  $V_0$ be a two-dimensional Artin representation  of a real quadratic field $F$ of conductor $\fD\subset \cO_F$ with
$\gcd(N,\fD)=1$    which is odd at the two real places of $F$,  and 
let 
\begin{equation}
 V = {\rm Ind} ^{\otimes} (V_0)
 \end{equation}
be the tensor induction of $V_0$  from $G_F$ to $G_\Q$.
If
\begin{equation}
\label{eqn:transfer1}
 {\rm Ver}_F^\Q \left(\det V_0\right) = 1,
 \end{equation}
then $V$ is self-dual and
 \begin{equation}
 {\rm sign}(E,V)= \varepsilon_F(N).
 \end{equation}
 }
 
\medskip
\noindent
 This strikingly simple recipe for  ${\rm sign}(E,V)$, which is identical to
 \eqref{eqn:sign-rcc},
suggests that 
$E(H)^V$ is non-trivial for an  explicit, systematic   class of Asai  representations, and it is natural to seek a construction  of the non-trivial
 elements of $E(H)^V$ whose existence is predicted by the equivariant Birch and Swinnerton-Dyer conjecture, guided  by the ideas in the proof of the Gross-Zagier formula.
Proposing such an analytic expression is one of the main goals of this paper.

\medskip
Assume from now on that $V_0$ and $V$ are as in the statement of the Sign Formula above, and that sign$(E,V)=-1$. 
It follows that $\varepsilon_F(N)=-1$, which implies that there is a prime $p$ dividing $N$ with odd multiplicity for which $\varepsilon_F(p)=-1$. Fix such a $p>3$ and assume for simplicity that it divides $N$ with multiplicity one. 

\bigskip
In fact,   the rest of the paper will focus solely  on 
the case where the conductor $N$ of $E$  satisfies the following somewhat stronger
{\em generalised Heegner 
hypothesis}:
\begin{equation}
\label{eqn:heegner-hyp}
N=pM,  \mbox{ where } \varepsilon_F(p) = -1,  \mbox{ and }   \varepsilon_F(\ell) = 1  \mbox{ for all } \ell |M.
\end{equation} 
 This hypothesis, although  stronger than  the more natural condition $\varepsilon_F(N) =-1$,   simplifies the setting without eliding any of its essential features. It implies that there is an ideal $\fM\subset \cO_F$ for which 
$\cO_F/\fM = \Z/M\Z$, which shall be fixed from now on. 

\bigskip
The Artin representation $V_0$ is associated to a
holomorphic Hilbert modular form over $F$ of parallel weight one, level $\fD$ and character $\chi:= \det V_0$. Such a Hilbert modular form is attached to a function
\begin{equation}\label{eqn:G-as-function-on-ideals}
G: I(F) \  \lra  \ \kappa
\end{equation}
on the semigroup $I(F)$ of integral ideals of $F$,  which is defined by extending by multiplicativity  the following definition on prime powers:
$$ G(\lambda) = {\rm Trace}\left(\sigma_{\lambda}|_{V_0^{I_\lambda}}\right),\qquad
G(\lambda^{n+1}) = \left\{
\begin{array}{ll} G(\lambda) G(\lambda^n) -  \chi(\lambda)G(\lambda^{n-1})  & \mbox{ if }  \lambda\nmid \fD,\\
G(\lambda)^n & \mbox{ if } \lambda| \fD, \end{array} \right.$$
where $I_\lambda$ is the inertia group at $\lambda$ and $V_0^{I_\lambda}$ is the space of $I_\lambda$-invariants.
Fix an ordering of the real embeddings of $F$ and for $\nu\in F$, write $\nu_1, \nu_2\in \R$ for the images of $\nu$ under these two embeddings.
To the function $G$ and to the  ideal $\fM$ of $\cO_F$  is associated a Hilbert modular generating series
\begin{equation}
\label{eqn:modular-gen-series} G_\fM(\tau_1,\tau_2) := \sum_{\nu\in (\fM\mathfrak{d}^{-1})_+} 
G((\nu) \mathfrak{d}) q^{\nu}, \qquad 
q^{\nu} := e^{2 \pi i (\nu_1 \tau_1 + \nu_2 \tau_2)},
\end{equation}
where the sum runs over all the totally positive elements of $\fM \mathfrak{d}^{-1}$, and $\mathfrak{d}$ denotes the different ideal of $\cO_F$.
This generating series is a Hilbert modular form of parallel weight one and character $\chi$
on the group
\begin{equation}
\label{eqn:Gamma0DM}
\Gamma_0(\fD;\fM) := \left\{ \mat{a}{b}{c}{d}:  \qquad a,d\in\cO_F,  \ \ \   b \in \fM^{-1}, \ \ \ c \in \fM\fD \ \right\}.
\end{equation}
After choosing an embedding of $F$ into a completion $F_p$ at $p$, and a $p$-adic logarithm $\log_p: F^\times \lra F_p$,
 we consider the following $p$-adic generating series deforming \eqref{eqn:modular-gen-series}:
\begin{equation}
\label{eqn:deformation-series} G_\fM'(\tau_1,\tau_2) := \sum_{\substack{\nu\in (\fM\mathfrak{d}^{-1})_+, \\ p\nmid \nu}} 
\log_p(\nu) G((\nu)\mathfrak{d})   q^{\nu}. 
\end{equation}
 Let 
 \begin{equation}\label{eqn:Phi}
 \tilde\Phi_{DM}^{(p)}:=  G_\fM'(\tau,\tau) =
 \sum_{\substack{\nu\in (\fM\mathfrak{d}^{-1})_+, \\ p\nmid \nu}} 
\log_p(\nu) G((\nu)\mathfrak{d})   q^{{\rm Tr}(\nu)}
   \end{equation}
 denote the diagonal restriction of $G_\fM'$, and let
 $$ 
 \Phi_{DM}^{(p)} := T \tilde\Phi_{DM}^{(p)},$$
 where $T$ is any  operator in the Hecke algebra over $\Q$ that annihilates  all classical modular forms of weight $2$ on $\Gamma_0(DM)$.
 Here $D$ is the positive  integer for which $\fD \cap \Z = (D)$.

 As explained in greater detail in Section \ref{sec:padicL}, the form 
  $\Phi_{DM}^{(p)} $ is the derivative with respect to the weight of a $p$-adic family of nearly overconvergent modular forms on $\Gamma_0(M)$  which specialises to $0$ in weight two. It is therefore
a    $p$-adic  modular form of weight two and (tame) level $DM$.  
 Let
\begin{equation}\label{eqn:PhiMp}
 \Phi_M^{(p)}:= {\rm Trace}^{DM}_M(\Phi_{DM}^{(p)})
 \end{equation}
be its  trace from level $DM$ to level $M$.
  Let $e_{\rm ord}$ be the ordinary projection from this space of  $p$-adic modular forms  to the space of classical modular forms of 
 weight two and level $N$ that are ordinary,
 and let $e_1$ be the further projection from this space to the subspace of modular forms which are fixed by $U_p^2$ (and hence new at $p$). 
 The modular form attached to the Asai representation $V$ is then defined  by
 \begin{equation}
 \label{eqn:def-Phi-N}
  \Phi_V :=   e_1 e_{{\rm ord}} (\Phi_{M}^{(p)}) \in M_2(\Gamma_0(N))^{{\rm new}-p}. 
    \end{equation}
It  is a classical modular form of weight two on
 $\Gamma_0(N)$ that is new at $p$. 
 
 Let $J_0(N)$ denote the Jacobian of $X_0(N)$ and define $$J_0(N)(H)^V = \hom_{G_\Q}(V, J_0(N)(H)\otimes \kappa)$$ as in \eqref{EV}.
The main conjecture of this paper relates $\Phi_V$ to   the $p$-adic logarithm of a global point arising from an element of $J_0(N)(H)^V$.

The $p$-new part of $J_0(N)$, denoted $J_0^{(p)}(N)$ has purely toric reduction, and there is a natural exact sequence arising from the Tate-Morikawa uniformisation of $J_0^{(p)}(N)$,
$$ 1 \lra X \stackrel{j}{\lra} T \lra J_0^{(p)}(N)(\bar\Q_p) \lra 1,$$
where $T$ is a $p$-adic torus, $X   := \hom(T,{\mathbb G}_m)$ is the character group of $T$ (a free $\Z$-module of rank
$g= \dim J_0^{(p)}(N)$), and $j$ arises from the monodromy pairing $X\times X\lra {\mathbb G}_m$.
A differential on $T$ is said to be 
{\em toric} if it is of the form $\eta^*(\frac{dt}{t})$  for some
$\eta\in X$. Such a differential is invariant under translation by $j(X)$, and theorefore  descends to a differential on   $J_0^{(p)}(N)$ . The space $\Omega^1_{\rm tor}$ of toric differentials on 
$J_0^{(p)}(N)$  is a free $\Z$-module of rank $g$  by definition, and is endowed with a natural linear action of the Hecke operators $T_n$ 
acting  as correspondences on
$J_0^{(p)}(N)$.  A toric differential is said to be {\em generic} if its translates under the Hecke operators $T_n$ span the $\Q$-vector space $\Omega^1_{\rm tor}\otimes \Q$.

\bigskip
\noindent
{\bf Main Conjecture}. {\em 
There is a point $P_V\in J_0^{(p)}(N)(H)\otimes \kappa $ belonging to the image of an element of $J_0^{(p)}(N)(H)^V$,  and a generic toric differential $\omega$
on $J_0^{(p)}(N)$  
for which
\begin{equation}
 \Phi_V = \sum_{n=1}^\infty \log_\omega(T_n(P_V)) q^n.
 \end{equation} }
  
  \bigskip  
  
 
 
 One of the simplest cases  of this conjecture is considered in \cite{DPV},
  where  $V_0$ is taken to be the direct sum of the trivial character  and an odd (unramified) ring class character $\psi$ of $F$, and $G$ is the Hilbert  Eisenstein series over $F$ of parallel weight one attached to this pair of  characters.
  The Asai representation of $V_0$  is then equal to 
  $$ {\rm Asai}(V_0) =: V =  1 \oplus 1 \oplus {\rm Ind}_F^\Q (\psi).$$
  The main result of \cite{DPV} 
   expresses  the $n$th fourier coefficients of the modular form $\Phi_V$ (where $\fM=1$ and $N=p$)
  as a linear combination of RM  values of the {\em winding cocycle} 
  \begin{equation}
  J_{w}\in H^1(\SL_2(\Z[1/p]),\cA^\times/\C_p^\times),
  \end{equation}
   where $\cA^\times$ is the multiplicative group of rigid analytic functions on the Drinfeld $p$-adic upper half-plane $\cH_p$. 
More precisely, the set of
RM points on $\SL_2(\Z[1/p])\backslash \cH_p$ of (prime-to-$p$)
discriminant $D_F={\rm Disc}(F)$ is endowed with an action of the class group of $F$, and the main result of loc.cit. is  the equality,\begin{equation}
 \label{eqn:dpv}
  a_n(\Phi_V) = \sum_{\sigma\in {\rm Gal}(H/F)} \psi(\sigma)^{-1} \log_p {\rm Nrm} J_w[T_n \tau^\sigma],
  \end{equation}
  where the value $J_w[\tau]$ of the cocycle $J_w$ at an RM point  has been extended to RM divisors by multiplicativity, and ${\rm Nrm}$ denotes the norm from $\Q_{p^2}$ to $\Q_p$.  The general theory of rigid analytic and meromorphic cocycles predicts that the RM values $J_w[T_n \tau^\sigma]$  map to   points in $J_0(p)(H)$ under the Tate-Morikawa uniformisation of $J_0(p)$. The Main Conjecture  of this paper can thus be envisaged as an extension of \eqref{eqn:dpv}
   to a setting where Hilbert Eisenstein series are replaced by general Hilbert modular forms of weight one (satisfying the self-duality assumption).
  
  
For an earlier discussion that also aims to place \eqref{eqn:dpv} in a more general framework, see
  \cite{pozzi-win}.
  
  \medskip
 

\section{Refinement of the main conjecture}\label{sec1}
 Write
 \begin{equation}
 \label{eqn:spectral-expansion}
{\Phi}_V = \sum_f {\lambda}_{V,f}  \cdot f + {\Phi}_V^{\rm old},
  \end{equation}
 where the sum is taken over a basis of normalised newforms in
 $S_2(\Gamma_0(N))$ and ${\Phi}_V^{\rm old}$ is an oldform.  Our goal is to reformulate the main conjecture of the introduction as a description of the coefficients ${\lambda}_{V,f}$ in this decomposition.

 
The  $\Q$-algebra ${\mathbb T}_N$ generated by the Hecke operators $T_m$ with $\gcd(m,N)=1$ acting  on  $S_2(\Gamma_0(N))$  is an \'etale algebra,  isomorphic to a product of totally real fields. Assume that $\kappa$ is large enough to split ${\mathbb T}_N$, i.e., that
  $$ {\mathbb T}_N \otimes \kappa \simeq \bigoplus _f   \kappa,$$
   where the sum  is taken over  pairs $(f,N_f)$ with $f$  a normalised   newform of level $N_f|N$ with coefficients in $\kappa$.
  The idempotents $\pi_f \in {\mathbb T}_N\otimes \kappa$ attached to this decomposition lead to a direct sum decomposition
  $$ J_0(N)(H)\otimes \kappa = \bigoplus_f J_0(N)(H)_f, $$
  where $J_0(N)(H)_f := 
  \pi_f(J_0(N)(H)\otimes \kappa)$.
    The ``Stark-Heegner point" $P_V$ of the Main Conjecture of the introduction
  is the  sum of its $f$-isotypic components:
 $$ P_V  \ :=\  \sum_f P_{V,f} + P_{V,{\rm old}}  \ \in \  J_0(N)  \otimes \kappa, \qquad \mbox{ where }  P_{V,f}:= \pi_f(P_V) \in J_0(N)(H)_f^V.$$
 The main conjecture of the introduction   can  be reformulated as follows:   

\begin{conjecture}
\label{conj:dlr}
The coefficient  
${\lambda}_{V,f}$ arising in the spectral expansion
 \eqref{eqn:spectral-expansion}
of $\Phi_V$  is given by
\begin{equation}
\label{eqn:decomp-PhiN}
 {\lambda}_{V,f} = \log_\omega(P_{V,f})\end{equation} 
% where $P_{V,f}\in J_0(N)(H)_f^V$. 
up to a multiplicative constant in $\kappa$.
   \end{conjecture}
  To further lighten the notations, we will  focus henceforth on the case where $f$ is a  weight two  newform of level $N$ with  {\em rational} fourier coefficients, i.e., a form that corresponds to 
 an elliptic curve $E/\Q$ 
 of conductor $N$. In that case, the field $\kappa$ can simply be taken to be the field of coefficients of the Artin representation $V_0$.
 
 
 The running assumption that $p$ divides $N$   implies that $E$ has multiplicative reduction at $p$, and hence 
 that
$$
a_p:=a_p(E)  = 
\begin{cases}
+1 & \mbox{ if } E \mbox{ has split multiplicative reduction at } p, \\
-1 & \mbox{ if } E \mbox{ has non-split multiplicative reduction at } p.
\end{cases}
$$


 The base change of $E$ to the quadratic unramified extension $\Q_{p^2}$ of $\Q_p$ admits a  rigid-analytic
uniformization according to Tate's theory. Namely,  there is a $p$-adic period $q_E \in p\Z_p$ and an isomorphism of rigid-analytic varieties over $\bar \Q_p$
\begin{equation}\label{Tate-unif}
\Phi_{\mathrm{Tate}}: \bar\Q_p^\times/q_E^{\Z} \,\stackrel{\sim}{\lra} \, E(\bar\Q_{p})
\end{equation}
which is defined over $\Q_p$ when $a_p=+1$ 
and over $\Q_{p^2}$ when $a_p= -1$.

Let $\log_p: \bar\Q_p^\times \lra \bar\Q_p$ denote the branch of the $p$-adic logarithm such that $\log_p(q_E)=0$, and define
$$
\logtate: E(\bar\Q_p) \lra \bar\Q_p, \quad \logtate(x) := \log_p \Phi^{-1}_{\mathrm{Tate}}(x).
$$ 
Note that $\logtate$ is defined only at primes of multiplicative reduction, and is different from that with respect to the canonical
differential on $E$ itself (denoted $\log_{E,p}$ in \cite{DLR1}).
 Up to a non-zero rational factor, we then have
 $$ \log_\omega(P_{V,f}) =  \logtate(P_{V,f}).
 $$
  
   The main goal of this section is  to further refine 
 Conjecture \ref{conj:dlr}
  by predicting   when the coordinate $\lambda_{V,f}$  in
  %% \eqref{eqn:decomp-PhiN} 
  \eqref{eqn:spectral-expansion}
    gives rise  to a non-trivial element of the Mordell-Weil group
    $J_0(N)(H)_f$. 
  More precisely, the global sign of the Hasse-Weil Artin  $L$-function $L(E,V,s)$  is $-1$ and hence this $L$-function vanishes to odd order. 
 If $\ord_{s=1}L(E,V,s)=1$, 
   then there is a unique irreducible constituent $W_1$ of $V = W_0 \oplus W_1$   for which
   $$ L(E,W_0,1)\ne 0, \qquad
   \ord_{s=1} L(E,W_1,s) = 1.$$
   
   
Letting $\alpha_G$ and $\beta_G=\alpha_G^{-1}$ denote the eigenvalues of the frobenius element at $p$ in $G_F$ acting on the Artin representation $V_0$ attached to $G$,   
Proposition \ref{prop:behaviour-asai} below asserts that the eigenvalues of $\sigma_p$ in $G_\Q$ 
on its tensor induction $V$  are $1,-1, \alpha_G$ and $\beta_G$. 
\begin{conjecture}\label{main-conj}
The scalar ${\lambda}_{V,f}$ is non-trivial if and only if the following two conditions are satisfied:
\begin{enumerate}
\item[(I)] $\ord_{s=1}L(E,V,s)=1$;
\item[(II)] The invariant $a_p\in \{\pm 1\}$ has multiplicity $1$ as an eigenvalue of the frobenius element $\sigma_p$ acting on $V$,  and its associated eigenvector belongs to $W_1$.
\end{enumerate}
When the above conditions hold, then $\lambda_{V,f} = \logtate (P_{V,f})$ and the global point $P_{V,f}$ lies in the image of $E(H)^{W_1}$. More precisely, this point forms a basis of the one dimensional subspace of this image on which $\sigma_p$ acts with the eigenvalue $a_p$.
\end{conjecture}
The remainder of this paper will be devoted to describing
  evidence for Conjecture \ref{main-conj}. Theorem \ref{thm:factorisation-padicL} provides theoretical evidence in the special case where $G$ is associated to the theta series of a character of a biquadratic field,  and 
 Section \ref{Sec-Experiments} documents a series of  numerical experiments 
 in support of this conjecture which also illuminate some of its more subtle features.

%\begin{remark}
%In generic scenarios, for instance where $V$ is irreducible, we expect the point $P_{V,f}$ to be non-trivial very often, although the question of how rare the vanishing of $P_{V,f}$ might be is poorly understood. In special scenarios, notably when $V$ is reducible, the %systematic vanishing of  
%$P_{V,f}$ may occur when condition (II) of
%Conjecture 
%\ref{main-conj} is not satisfied: see for example the analysis performed in \S \ref{sec:CM} when $G$ is a theta series.
%\end{remark}

%\medskip

\section{Asai representations}
\label{sec:asai-rep}
This section collects a few simple facts about Asai representations, in the level of generality that is relevant to the constructions of this article.

Let $\fG$ be a finite group and let  $G$ be a subgroup of index two in $\fG$.
The induction to $\fG$ of
a finite-dimensional representation  $V_0$ of $G$ is the space of $G$-invariant functions
$$ {\rm Ind}_G^{\fG}(V_0)  := \{ \eta:\fG\lra V_0 \mbox{ satisfying }
\eta(g x) = g \eta(x),  \mbox{ for all } g\in G \} $$
endowed with the action of $\gamma\in \fG$ given by
$$ (\gamma\cdot \eta)(x) =  \eta(x\gamma).$$
Since a vector in ${\rm Ind}_F^\Q(V_0)$ is completely determined by its values on 
a system of coset representatives for $G\backslash\fG$,
the  representation ${\rm Ind}_G^{\fG}(V_0)$
is identified with $V_0 \oplus V_0$  after 
fixing an element $\tau_0 \in \fG-G$, and sending $\eta$ to
$(\eta(1), \eta(\tau_0))$. 
  Conjugation by this element $\tau_0$  determines an automorphism of $G$  which will be written as $g\mapsto g':= \tau_0 g \tau_0^{-1}$.
Under the
 identification of ${\rm Ind}_G^{\fG}(V_0)$ with $V_0\oplus V_0$ determined by 
$\tau_0$, the action of 
$\fG$ on  the latter space is given by
\begin{equation}
\label{eqn:def-usual-ind}
 g((v_1, v_2)) =  
\left\{ \begin{array}{ll}
 (g v_1,  g' v_2) & \mbox{ if } g \in G, \\
 (g\tau_0^{-1} v_2, 
 \tau_0 g v_1) & \mbox{ if } g\in
  \fG -G.
 \end{array} \right.
\end{equation}
The
 {\em tensor induction}
of $V_0$ to $\fG$, denoted
${\rm Ind}^\otimes(V_0)$,  is defined by replacing direct sums with tensor products in  
\eqref{eqn:def-usual-ind}.
Its underlying vector space is $V := V_0 \otimes V_0$, with action of $\fG$ defined by  
\begin{equation}
\label{eqn:def-tensor-ind}
 g(v_1 \otimes v_2) =  
\left\{ \begin{array}{ll}
g v_1 \otimes  g' v_2 & \mbox{ if } g \in G, \\
 g\tau_0^{-1} v_2 \otimes \tau_0 g  v_1 & \mbox{ if } g\in
  \fG -G.
 \end{array} \right.
\end{equation}
 \begin{lemma} The character 
 $\chi_{_V}$ attached to $V = {\rm Ind}^\otimes(V_0)$ is given by  $$ \chi_{_V}(g) = \left\{
 \begin{array}{ll} 
 \chi_{_{V_0}}(g) \chi_{_{V_0}}(g'),
 & \mbox{ if } g\in G,
 \\
 \chi_{V_0}(g^2) & \mbox{ if } g\in \fG -G.
 \end{array} \right.$$
 \end{lemma}
\begin{proof}
For $g\in G$, the formula for $\chi_{_V}(g)$  follows directly  from 
\eqref{eqn:def-tensor-ind}.
Consider the second case, where $g$ belongs to $ \fG-G$, and assume for notational simplicity that $V_0$ is two-dimensional, although the general case can be treated in the same way. Since the isomorphism class of the Asai induction does not depend on the choice of $\tau_0 \in\fG -G$
that was made to define it, we may set $\tau_0=g$. Every element of $G$ acts semisimply on $V_0$ 
since $G$ is finite; let 
$v_\alpha, v_\beta \in V_0$ be an eigenbasis for $g^2$ acting on $V_0$, with eigenvalues $\alpha$ and $\beta$ respectively.
Relative to the basis $(v_\alpha \otimes v_\alpha, v_\beta\otimes v_\beta, v_\alpha \otimes v_\beta, v_\beta\otimes v_\alpha)$ for $V$, the action of $g$ 
is given by
\begin{eqnarray*}
 g (v_\alpha \otimes v_\alpha) = \alpha \cdot  v_\alpha \otimes v_\alpha,  &&
\qquad 
g (v_\beta \otimes v_\beta)   = \beta \cdot v_\beta \otimes v_\beta,
\\
g (v_\alpha\otimes v_\beta) = \alpha \cdot v_\beta\otimes v_\alpha,    && \qquad
g (v_\beta\otimes v_\alpha) = \beta \cdot v_\alpha\otimes v_\beta. \end{eqnarray*}
The matrix of $g$ in this basis is therefore given by
\begin{equation}
\label{eqn:matrix-reflection}
 M_g:=  \left(\begin{array}{cccc}
\alpha & 0 & 0 & 0 \\
0 & \beta & 0 & 0 \\
0& 0& 0 & \beta \\
0 & 0 & \alpha & 0
\end{array}\right).
\end{equation}
The result follows directly.
\end{proof}

  If $V_0$ is a one-dimensional
representation of $G$, associated to a character $\chi:G\lra \C^\times$, the tensor induction of $V_0$ coincides with the  transfer of $\chi$.

\medskip
Of special interest for the constructions of this article are two-dimensional representations $V_0$ of $G$ satisfying the condition
\begin{equation}
\label{eqn:self-dual}
 {\rm Ind}^\otimes \det(V_0) = 1.
 \end{equation}
 \begin{proposition}
 \label{prop:behaviour-asai}
 If $V_0$ is a two dimensional representation of $G$ satisfying 
 \eqref{eqn:self-dual}, 
 then $V:= {\rm Ind}^\otimes V_0$ is a self-dual Artin representation. The eigenvalues of any $g\in \fG-G$ acting on $V$ are
 equal to $1$, $-1$, 
 $\alpha$,  and $\beta = \alpha^{-1}$,
 where $\alpha$ and $\beta$ are the eigenvalues of  $g^2\in G$  acting  on $V_0$. 
 \end{proposition}
\begin{proof}
If $g$ belongs to  $G$, let $\alpha$ and $\beta$ be the eigenvalues of $g$ acting on $V_0$, and let $\alpha'$ and $\beta'$ denote the eigenvalues of $g'$. Assumption
\eqref{eqn:self-dual} implies that $\alpha\beta\alpha'\beta'=1$, while the eigenvalues for $g$ on $V$ are equal to $\alpha\alpha'$, $\alpha \beta'$, 
$\beta\alpha'$, and $\beta\beta'$.
This set of eigenvalues is therefore closed under the map $\zeta \mapsto \bar\zeta$, and hence the trace of $g$ acting on $V$ is real. 
If $g$ belongs to $\fG -G$, then letting $\alpha$ and $\beta$ denote the eigenvalues of $g^2$ acting on $V_0$, condition
\eqref{eqn:self-dual} implies that 
\begin{equation}
\label{eqn:alphabeta1}
\alpha\beta= \det(V_0)(g^2) = {\rm Ind}^\otimes \det(V_0)(g) = 1.
\end{equation}
 Since the 
matrix of $g$ acting on $V$ relative to a suitable basis  is given by
\eqref{eqn:matrix-reflection}, 
its trace $\alpha+\beta$ is real, and the self-duality of $V$  follows. The second assertion in Proposition \ref{prop:behaviour-asai}
 follows likewise from
\eqref{eqn:matrix-reflection}  and
\eqref{eqn:alphabeta1}.
\end{proof}


\section{Reinterpretation via $p$-adic $L$-functions}
\label{sec:padicL}


The purpose of this section is to relate the conjecturally global point $P_{V,f}$ of Conjecture \ref{main-conj}
to derivatives of suitable $p$-adic $L$-functions. This will lead to  some theoretical evidence for Conjecture \ref{main-conj} when $V_0$ is induced from a character of a quadratic extension of $F$ which is biquadratic over $\Q$ (Theorem \ref{thm:factorisation-padicL}).
 
Following the notations of  earlier sections,
recall that 
 $G$  is the holomorphic Hilbert eigenform of conductor $\fD$, nebentype $\chi$ and parallel weight $1$ over a real quadratic field $F$  whose Asai representation is isomorphic to $V$, and that
 $f\in S_2(N)$ is a normalised newform  associated to an elliptic curve $E$ of conductor $N=pM$ by modularity. Here as above $p>3$ is a prime and $M\geq 1$ an integer relatively prime to $p$. It is assumed that $(N, D_F \fD)=1$ and $(F,N)$ 
 satisfy the generalised Heegner hypothesis of 
 \eqref{eqn:heegner-hyp}.
 
 Because $U_pf = \pm f$, the weight two newform $f$ is ordinary at $p$. It can therefore be realized as the weight two specialization   of a   {\em Hida family} of modular forms, denoted $\hf$. This Hida family is a formal $q$-expansion with coefficients in a finite flat
  extension $\Lambda_{\hf}$ of the Iwasawa algebra $\Lambda :=\Z_p[[(1+p\Z_p)^\times ]]
\simeq \Z_p[[T]]$,
 as defined in \cite[Def.\,2.16]{DR1}, whose specialisations at a dense set of {\em classical} points are classical eigenforms of tame level $N$ and varying weights. 
 More precisely, the
   spaces
 $$\cX_{\hf} :=
 \hom_{\rm alg}(\Lambda_{\hf}, \C_p),    \qquad 
 \cX :=  \hom_{\rm alg}(\Lambda, \C_p)  = 
 \hom_{\rm grp}((1+p\Z_p)^\times, \C_p^\times)$$
 are endowed with a natural structure of  $p$-adic analytic spaces. The {\em weight space} $\cX$ contains the integers as a countable dense subset via the inclusion $k \mapsto (x\mapsto x^{k-2})$, and restriction gives a   finite flat morphism
 $$  \w: \cX_{\hf}  \lra 
\cX,$$
called the {\em weight map}.  By definition, the specialisation $f_x:= x(\hf)$ of $\hf$ at
a point $x\in \cX_{\hf}$ is an ordinary overconvergent  $p$-adic eigenform of level $N$ and weight $k:= \w(x)$.
 When $k:=\w(x)$ belongs to $\Z^{>2}$, Hida's classicality theorem implies that   $f_x$ is a classical modular form of weight $k$ on $\Gamma_0(N)$ which is new at the primes dividing $M$ and is a simultaneous eigenvector for the Hecke operators, satisfying
$$ U_p f_x = \alpha_x f_x, \quad \mbox{ with }  \alpha_x \in \cO_{\C_p}^\times.$$
The Hida family $\hf$ is uniquely characterised by 
these properties along  with the condition that $f_{x_0} = f$, where $x_0\in \cX_{\hf}$ is a suitable classical point of weight $\w(x_0)=2$.


The ordinarity of $f_x$ when $\w(x) >2$ 
 implies that  $f_x$ cannot be new at $p$: rather, it is the ordinary $p$-stabilisation of a classical   normalised newform of level $M$,
denoted  $f_x^\circ$.


Let $L(f_x^\circ,V,s)$ denote the Hecke-Artin $L$-series attached to $f_x^\circ$ and to $V$, i.e.,  to the tensor product 
of $V$ with the compatible system of two-dimensional Galois representations attached to $f_x^\circ$. A twisted variant of Rankin's method, which will be detailed below, implies that $L(f_x^\circ,V,s)$ admits analytic continuation and a functional equation relating its values at $s$ and $k-s$. Denoting by ${\rm sign}(f_x,V)$ the sign that arises in this functional equation, 
the generalised Heegner hypothesis \eqref{eqn:heegner-hyp} implies that
$$ {\rm sign}(f_x,V) = 
\left\{ \begin{array}{rl}
 -1 & \mbox{ if } x=x_0;\\
 1 & \mbox{ if } k:=\w(x)>2.
 \end{array} \right.$$ 
In particular,  the central critical  values   $L(f_x^\circ,V, k/2)$ need not vanish when $k >2$.

It is therefore natural to attempt to interpolate these  values $p$-adically to a function on $\cX_{\hf}$. Starting with the fundamental work of Ichino \cite{Ich} and following the principles of \cite{DR1}, this task has been undertaken recently in the literature in various degrees of generality thanks to the contributions of Hsieh \cite{Hs}, Blanco-Fornea \cite{BF}, Chen-Cheng \cite{CC}, Ishikawa \cite{Ish}, Kazi \cite{Kazi} and Kazi-Loeffler \cite{KL}.

In order to describe the results, recall the Hilbert  generating series $G_{\fM}(\tau_1,\tau_2)$ attached to $G$ and $\fM$ in \eqref{eqn:modular-gen-series}.  Let 
$$ \delta_{k_1} = 
\frac{1}{2\pi i} \left(\frac{d}{d\tau_1} - i \frac{k_1}{2y_1}\right) $$
be the partial Shimura-Maass derivative operator mapping holomorphic Hilbert modular forms of weight $(k_1,k_2)$ to  {\em nearly holomorphic}
Hilbert modular forms of weight $(k_1+2,k_2)$, and write
$$\delta_{k_1}^t := \delta_{k_1+2t-2} \circ \cdots \circ \delta_{k_1+2} \circ \delta_{k_1}$$
for its $t$-fold iterate.
The nearly holomorphic modular form
$$ \delta_{1}^t G_{\fM}(\tau_1,\tau_2) \in M_{1+2t,1}^{\rm nh}(\Gamma_0(\fD,\fM))$$
transforms like a holomorphic Hilbert modular form of weight $(1+2t,1)$ under the action of the congruence subgroup of \eqref{eqn:Gamma0DM}. For any even weight $k = 2t+2>2$, 
the diagonal restriction
$$ H_k^{\rm nh}(q) := \delta_1^t G_\fM(\tau,\tau) \in M_k^{\rm nh}(\Gamma_0(DM))
$$ 
is a nearly holomorphic modular form of weight $k$ and  level $DM$. 
Let 
\begin{equation}
\label{eqn:def-Hk}
H_k:=   e_{\rm hol}  {\rm Tr}^{DM}_M H_k^{\rm nh} \in M_k(\Gamma_0(M))
\end{equation}
denote  the holomorphic projection of its trace to level $M$.
Then  for all $x\in \cX_f$ with $\w(x)=k>2$, let 
\begin{equation}
\label{eqn:triple-period-C}
I(f_x^\circ, G)  :=  \frac{\langle H_k, f_x^\circ\rangle}{\langle f_x^\circ,f_x^\circ\rangle},\end{equation}
where $\langle \ , \ \rangle$ denotes the Petersson scalar product on   weight $k$ modular forms.

\vspace{0.3cm}

Let $d_1 := q_1 \frac{d}{dq_1}$ be the partial $d$ operator   which maps Hilbert modular forms of weight $(k_1,k_2)$ to nearly overconvergent Hilbert modular forms of weight $(k_1+2,k_2)$. The generating
series
\begin{equation}
\label{eqn:theta-G} 
d_1^t G_\fM(q_1,q_2) := \sum_{\nu\in (\fM\mathfrak{d}^{-1})_+} 
\nu_1^t \cdot G((\nu)\mathfrak{d})   q_1^{\nu_1} q_2^{\nu_2} \ 
\in \ M_{1+2t,1}^{\rm no}(\Gamma_0(\fD,\fM))
\end{equation}
is the $q$-expansion of a {\em nearly overconvergent} $p$-adic Hilbert modular form of weight $(1+2t,1)$.   
Its diagonal restriction  
$$ H_k^{\rm no} := d_1^t G_\fM(q,q) \in M_k^{\rm no}(\Gamma_0(DM))$$ 
is a nearly overconvergent modular form of weight $k$ and tame level $DM$. Let 
$$ H_k^\flat :=  e_{\rm ord}  {\rm Tr}^{DM}_M H_k^{\rm no} \in M_k(\Gamma_0(N))$$
be the  classical modular form of weight $k$ and level $N$ obtained by applying to $H_k^{\rm no}$:
\begin{enumerate}
\item
the trace map ${\rm Tr}^{DM}_M$ from overconvergent forms of  level $DM$ to level $M$;
\item the ordinary projection $e_{\rm ord}$ from overconvergent modular forms of tame level $M$ to classical ordinary modular forms of level $N$.
\end{enumerate} 
The geometric principles evoked in the proof of
\cite[Prop.~2.8]{DR1}
imply that
$$ H_k^\flat = e_{\rm ord}H_k,$$
and hence $H_k^\flat$ 
can be envisaged as a $p$-adic avatar of the modular form $H_k$
of \eqref{eqn:def-Hk}. 
Replacing $H_k$ by $H_k^\flat$ in
\eqref{eqn:triple-period-C}
leads to  an alternate  expression for $I(f_x^\circ,G)$ in terms of the period $\langle H_k^\flat, f_x\rangle  \langle f_x,f_x\rangle^{-1}$.

Although more genuinely $p$-adic in nature, 
this quantity does not interpolate  $p$-adically to an analytic or even continuous function of $x\in \cX_{\hf}$, because the quantity $
 \nu^t$ arising in the fourier coefficients of $d_1^t G_{\fM}$ are only analytic functions of $t$ when 
 $p\nmid \nu$.
 This motivates replacing the form
 $d_1^t G_\fM(q_1,q_2)$ by its
 $p$-depletion in 
 \eqref{eqn:theta-G}: \begin{equation}
\label{eqn:theta-G-bis} 
d_1^t G_\fM^{[p]}(q_1,q_2) := \sum_{\substack{\nu\in (\fM\mathfrak{d}^{-1})_+\\ p\nmid \nu}} 
\nu_1^t G((\nu)\mathfrak{d})   q_1^{\nu_1} q_2^{\nu_2},
\end{equation}
a modular generating series whose fourier coefficients interpolate to  $p$-adic analytic functions of $t\in (\Z/(p^2-1)\Z) \times\Z_p$.
Let 
$$H_k^{[p]}  := e_{\rm ord}  {\rm Tr}^{DM}_M \left(d_1^t G_\fM^{[p]}(q,q) \right)  \in M_k(\Gamma_0(N))$$ 
be the 
 ordinary projection of the trace to level $M$ of the diagonal restriction of this nearly overconvergent Hilbert modular form,
  and set
 \begin{equation}
 \label{Lp-def} 
 \Lp(\hf,V)(x) := 
 \frac{\langle H_k^{[p]}, f_x\rangle}{\langle f_x,f_x\rangle}, \qquad k:= {\rm w}(x). 
 \end{equation}
 This quantity interpolates to a $p$-adic meromorphic function of $x$, denoted $\Lp(\hf,V)$, and commonly referred to as the {\em twisted triple-product $p$-adic $L$-function}. Since it suffices for our purposes, here we have limited ourselves to introduce this $p$-adic $L$-function as a single-variable function on the weight variable of $\hf$, but the reader is invited to consult the references above for a three-variable version of it.

%After specialising \cite[Theorem 1.3]{BF} 
%to the ``weight $(k,1,1)$ line", we obtain
% $$ \Lp(\hf,V)(x) = 
% \cE(f_x,V) \cdot \cE_0(f_x)^{-1} \cdot \frac{\langle H_k, f_x^\circ\rangle}{\langle f_x^\circ, f_x^\circ\rangle},$$
% and Theorem \ref{thm:padicL} follows.


As anticipated before, $ \Lp(\hf,V)$ interpolates the square-root of the algebraic part of central critical classical $L$-values, which we now describe more precisely. Here we state a version of this result proved by Ishikawa in \cite[Theorem 1.5.1]{Ish}  although the above references show it holds in greater generality and under more relaxed assumptions. 

Recall that $\alpha_G$ and $\beta_G$ are the eigenvalues of the frobenius element at $p$ in $G_F$ acting on the Artin representation $V_0$ attached to $G$, and
the eigenvalues of $\sigma_p$ in $G_\Q$ 
 on its tensor induction $V$  are $1,-1, \alpha_G$ and $\beta_G$, by Proposition
 \ref{prop:behaviour-asai}. 
 
 Given two functions $A(x)$ and $B(x)$  on a subset of classical points of $\cX_{\hf}$ that is dense for the rigid-analytic topology,  we will write
\begin{equation}
\label{eqn:sim}
 A(x) \sim B(x)
 \end{equation}
if they differ by a quantity that interpolates to a $p$-adic meromorphic function on
$\cX_{\hf}$ that is regular at $x_0$ and whose value  at $x_0$   belongs to 
$\kappa^\times$. We call such functions {\em admissible}.


\begin{theorem}[Ishikawa]
 \label{thm:padicL}
Assume 
\begin{enumerate}
\item[($H_p$)] The $\mbox{mod } p$ residual Galois representation $\bar{\varrho}_f: G_{\Q}\lra \, \GL_2(\mathbb{F}_p)$ is absolutely irreducible and the restriction of $\bar{\varrho}_f$ to the decomposition group at $p$ is the sum of two {\em distinct} characters.

\item[($H_M$)] The local admissible representation associated to $f$ at a prime $q\mid M$ is not supercuspidal (which is automatically fulfilled when $q\mid \mid M$).

\end{enumerate}

For all  classical points
$x\in \cX_f$  of weight $k =2t+2$ with $t\geq 0$ we have 
$$
 \Lp(\hf, V)(x)^2 \, \sim \, \frac{\mathcal{E}(f_x,V)}{ \mathcal{E}_0(f_x)} L_{\rm alg}(f_x^\circ,V,k/2)
 $$
where 
\begin{equation}\label{Euler-Asai}
\mathcal{E}(f_x,V) =  \left(1 -  \frac{p^{t}\alpha_G}{\alpha_x }\right)^2 \left(1 -  \frac{p^{t}\beta_G}{\alpha_{x} }\right)^2, 
\qquad
\cE_0(f_x) =  \left(1 - \frac{p^{k-1}}{\alpha^2_{x}}\right)^2
\end{equation}  
and
\begin{equation}\label{Lalg}
L_{\rm alg}(f_x^\circ,V,k/2) =  \frac{  (\frac{k-2}{2}!)^4  } { \pi^{2k}
\langle f_x^\circ, f_x^\circ\rangle^2}   L(f_x^\circ,V,k/2).
\end{equation}
\end{theorem}

\begin{proof} This follows from \cite[Theorem 1.5.1]{Ish}. Indeed, observe first that all assumptions in loc.\,cit.\,are fulfilled, since our running hypothesis that $(N, D_F \fD)=1$ implies Ishikawa's set $\Sigma^-$ is empty. The formula in  loc.\,cit.\,asserts that
\begin{equation*}\label{Ishi-formula}
 \Lp(\hf, V)(x)^2 =   \Gamma_{f_x,g}(0) \frac{L(f_x^\circ,V,k/2)}{(-1)^{\alpha_g} D^t  \,\Omega_{f_x}^2}   \mathcal{E}^{\dagger}(f_x,g)  \prod_{\ell\in \Sigma^{\text{dist}}} (1+\ell^{-1})^2 \prod_{\ell\in \Sigma^{\text{excep}}} ((\ell+1)^2-\alpha^2_{x})^2
\end{equation*}
where
\begin{enumerate}
\item[(i)] $\Gamma_{f_x,g}(s)$ is the Gamma function introduced in  \cite[(1.2.1)]{Ish}, whose value at $s=0$ is $ (\frac{k-2}{2}!)^4   \pi^{-2k} $ up to a power of $2$ (whose exponent is a polynomial in $k$), and thus interpolates to an admissible function;
\item[(ii)] $\Sigma^{\text{dist}}$ and $\Sigma^{\text{excep}}$ are finite sets of primes distinct from $p$ introduced in  \cite[\S 1.5]{Ish}; note that the functions    $((\ell+1)^2-\alpha^2_{x})^2$  interpolate to admissible $p$-adic analytic functions on $\cX_{\hf}$;


\item[(iii)] the period $\Omega_{f_x}$ is defined at the end of \cite[\S 1.4]{Ish} and is precisely the Petersson scalar product $\langle f_x^\circ, f_x^\circ\rangle$ quoted above, up to an Euler-like factor $\mathcal{E}_p(f_x,\mbox{Ad})$ introduced in loc.cit, a power of $2$ that again interpolates to an admissible function of $k$, and a function denoted $x(\eta_{\hf})$ given by a choice of generator of the congruence ideal of $f_x$, which  also interpolates to an admissible $p$-adic analytic function on $\cX_{\hf}$. 
\end{enumerate}

Since the ratio $  \mathcal{E}^{\dagger}(f_x,g)/ \mathcal{E}_p(f_x,\mbox{Ad})^2$ is readily seen to be equal to the ratio $\mathcal{E}(f_x,V)/ \mathcal{E}_0(f_x)$, it follows that Ishikawa's formula quoted above is equal to the one recorded in the main statement of the theorem after removing quantities that visibly interpolate to an admissible $p$-adic meromorphic function on
$\cX_{\hf}$. 
\end{proof}


%\vspace{1cm}

%Rather than interpolating 
%the central critical values 
%$L(f_x^\circ,V,k/2)$ themselves, we will first evoke an integral representation of these   values (divided by a suitable period) that more readily lends  itself  to $p$-adic interpolation.

%Recall the Hilbert  generating series $G_{\fM}(\tau_1,\tau_2)$ attached to $G$ and $\fM$ in \eqref{eqn:modular-gen-series}.  Let 
%$$ \delta_{k_1} = 
%\frac{1}{2\pi i} \left(\frac{d}{d\tau_1} - i \frac{k_1}{2y_1}\right) $$
%be the partial Shimura-Maass derivative operator mapping holomorphic Hilbert modular forms of weight $(k_1,k_2)$ to  {\em nearly holomorphic}
%Hilbert modular forms of weight $(k_1+2,k_2)$, and write
%$$\delta_{k_1}^t := \delta_{k_1+2t-2} \circ \cdots \circ \delta_{k_1+2} \circ \delta_{k_1}$$
%for its $t$-fold iterate.
%The nearly holomorphic modular form
%$$ \delta_1^t G_{\fM}(\tau_1,\tau_2) \in M_{1+2t,1}^{\rm nh}(\Gamma_0(\fD,\fM))$$
%transforms like a holomorphic Hilbert modular form of weight $(1+2t,1)$ under the action of the congruence subgroup of \eqref{eqn:Gamma0DM}. For any even weight $k = 2t+2>2$, 
%the diagonal restriction
%$$ H_k^{\rm nh}(q) := \delta_1^t G_\fM(\tau,\tau) \in M_k^{\rm nh}(\Gamma_0(DM))
%$$ 
% is a nearly holomorphic modular form of weight $k$ and  level $DM$. 
%Let 
%\begin{equation}
%\label{eqn:def-Hk}
%H_k:=   e_{\rm hol}  {\rm Tr}^{DM}_M H_k^{\rm nh} \in M_k(\Gamma_0(M))
%\end{equation}
%denote  the holomorphic projection of its trace to level $M$.
%Then  for all $x\in \cX_f$ with $\w(x)=k>2$, let 
%\begin{equation}
%\label{eqn:triple-period-C}
%I(f_x^\circ, G)  :=  \frac{\langle H_k, f_x^\circ\rangle}{\langle f_x^\circ,f_x^\circ\rangle},\end{equation}
%where $\langle \ , \ \rangle$ denotes the Petersson scalar product on   weight $k$ modular forms.


%Write $\mathfrak{D} = \mathfrak{D}^+\mathfrak{D}^-$ where all prime factors of $\mathfrak{D}^+$ (resp.\,of $\mathfrak{D}^-$) lie above primes that split (resp.\,ramify or remain inert) in $F$.
%Given  two functions $A(x)$ and $B(x)$  on the classical points of $\cX_{\hf}$,  we will write
%\begin{equation}
%\label{eqn:sim}
% A(x) \sim B(x)
% \end{equation}
%if they differ by a quantity that interpolates to a $p$-adic meromorphic function on
%$\cX_{\hf}$ that is regular at $x_0$ and whose value  at $x_0$   belongs to 
%$\kappa^\times$, as in the following statement.

%\begin{proposition}[Ichino, Hsieh, Chen-Cheng, Ishikawa]
%\label{prop:ichino}  Assume  $\mathfrak{D}^-$ is square-free.
%For all  classical $x\in \cX_f$  with $\w(x)>2$,
%the quantity $I(f_x^\circ,G)$ is algebraic and belongs to the field generated by $\kappa$ and by the fourier coefficients  of $f_x$. Furthermore,
%$$  I(f_x^\circ,G) ^2 \sim  L_{\rm alg}(f_x^\circ, V,k/2), $$
%where  
%$$
%L_{\rm alg}(f_x^\circ,V,k/2) = 
%\frac{(\frac{k-2}{2}!)^4   D^{\frac{k-1}{2}} L(f_x^\circ,V,k/2)} { (2\pi)^{2k}
%\langle f_x^\circ, f_x^\circ\rangle^2}.$$
% \end{proposition}
%\begin{proof} It follows from \eqref{eqn:triple-period-C} that $I(f_x^\circ,G)$ is the coefficient of $H_k$ at $f_x^\circ$ in the spectral decomposition of $M_k(\Gamma_0(M))$. The Hilbert modular form $G$, and thus also $H_k$, is defined over $%\kappa$, while $f_x^\circ$ is rational over the field generated by its Fourier coefficients. Hence it follows that $I(f_x^\circ,G)$ is algebraic and belongs to the composition of the two latter fields.

%Let $\Sigma_\Q$ and $\Sigma_F$ denote the set of places of $\Q$ and $F$ respectively and $$\pi_f \simeq \otimes'_{v\in \Sigma_\Q} \pi_{f,v}, \quad \pi_G \simeq \otimes'_{w\in \Sigma_F} \pi_{G,w}$$ denote the cuspidal automorphic %representations associated to $f$ and $G$. Let  $$\Pi = \pi_f \boxtimes \pi_G \simeq \otimes_{v\in \Sigma_{\Q}}' \, \pi_{f,v} \boxtimes \pi_{G,v}$$ denote the cuspidal automorphic representation of $\GL_2(\A_{\Q}) \times \GL_2(\A_{F})$ %associated to $f$ and $G$, where for every place $v\in \Sigma_\Q$ we set $\pi_{G,v} = \otimes_{w|v} \pi_{G,w}$. Ichino introduced in \cite{Ich} local and global pairings, that we invoke below following  the normalizations adopted by Ishikawa in %\cite[\S 2,p.\,624-625]{Ish}. Namely  a global $\GL_2(\A_{\Q}) \times \GL_2(\A_{F})$-invariant integral pairing
%$$
%I: \Pi \times \Pi^\vee \lra \C
%$$
%as well as local integral $\GL_2(\Q_v) \times \prod_{w|v}\GL_2(F_w)$-invariant pairings
%$$
%I_{_v}: \Pi_v \times \Pi_v^\vee \lra \C
%$$
%for every place $v\in \Sigma_{\Q}$; cf.\,\S 2 of loc.\,cit.\,for details. It follows from Ichino's main theorem 1.1 in \cite{Ich} (cf.\,\cite[Theorem 2.1]{Ish} for the formulation in our scenario) that for any $\phi = \otimes \phi_v\in \Pi$, $\phi' = \otimes %\phi'_v\in \Pi^\vee$:\footnote{Vic: I am denoting $D=\disc(F)$ the discriminant of $F$ because we have already introduced this notation. But to my taste it overlaps a bit with $\mathfrak{D}$ which is being used for the level of $G$. I think I prefer to denote the latter $\mathfrak{N}_G$. If you agree I will perform this notational change.}
%\begin{equation}\label{Ichino-formula}
%I(\phi,\phi') = \frac{2 \sqrt{D}}{\tilde\zeta_{\Q}(2)^2} \cdot  \tilde{L}(f_x^\circ,V,, k/2) \cdot \proD I_v(\phi_v,\phi'_v)
%\end{equation}
%where $\tilde\zeta_{\Q}$ is the completed Riemann zeta function and  $\tilde{L}(f_x^\circ,V,s) = 2^4 (2\pi)^{-4s} \Gamma(s)^4 L(f_x^\circ,V, s)$ is the completed $L$-function, which satisfy\footnote{Vic: Verify the powers of $\pi$ match with the %statement.}
%$$
%\tilde \zeta_{\Q}(2) = \pi^{-1} \zeta_{\Q}(2) = \frac{\pi}{6}, \quad \tilde{L}(f_x^\circ,V,k/2) = \frac{( 2 \cdot (\frac{k-2}{2})!)^4}{ (2\pi)^{2k}}  L(f_x^\circ,V, k/2).
%$$

%The product in \eqref{Ichino-formula} is well-defined because $I_v(\phi_v,\phi'_v)=1$ at all finite places of $\Q$ corresponding to primes that do not divide $M \mathfrak{D}$, as shown in \cite[Lemma 2.2]{Ich}. 

%At the place at infinity, $I_\infty(\phi_\infty,\phi'_\infty)$ was computed in \cite[Proposition 3.11]{CC} and shown to be a power of $2$ depending only on the weight of $f_x^\circ$.

%At primes $q\mid M \mathfrak{D}^+$, our running Heegner assumption implies that $q$ splits in $F$ and thus $(\Q \times F)_q \simeq \Q_q \times \Q_q \times \Q_q$. In this scenario Hsieh constructed in \cite{Hs} a pair of local test vectors $%(\phi_q,\phi_q') \in \Pi_q \times \Pi_q^\vee$ for which he explicitly computed the local Ichino integral $I_q(\phi_q,\phi'_q)$, showing in particular that these quantities interpolate to an analytic function on $\cX_{\hf}$ whose value at $x_0$ belongs to $\kappa^\times$. We refer to \cite[\S 5-6]{Hs} for details. 

%Besides, at primes $q\mid \mathfrak{D}^-$  the local Ichino integral $I_q(\phi_q,\phi'_q)$ was computed by Chen-Cheng \cite[Propositions 4.7 and 4.8]{CC} for a specific choice of local test vectors $(\phi_q,\phi_q') \in \Pi_q \times \Pi_q^\vee$. %Again it readily follows from their results that $I_q(\phi_q,\phi'_q)$ interpolate to an analytic function on $\cX_{\hf}$ whose value at $x_0$ lies in $\kappa^\times$.

%As in the proof of \cite[Theorem 7.4]{Ish}, 
%The global test vectors $\phi = \otimes \phi_v\in \Pi$, $\phi' = \otimes \phi'_v$ obtained by Ichino's and Chen-Cheng's recipes at ramified primes satisfy $I(\phi,\phi') \sim   \langle H_k, f_x^\circ\rangle^2$ where $H_k$ is constructed as above {\em %for a suitable $\fM$-stabilization} $G^\star$ of $G$, and the proposition follows.\footnote{Are you sure Henri that we can prove this proposition using the natural $\fM$-stabilization described in  \eqref{eqn:modular-gen-series}? Hsieh's choice of %test vectors is slightly more complicated, but it works well for our purposes. Of course the difference between one choice and another is expected to be a simple algebraic factor...}
%\end{proof}

 
% Because of Proposition
% \ref{prop:ichino}, it is natural to attempt  the   $p$-adic interpolation of the period $I(f_x^\circ,G)$ itself as 
%$x$ varies over $\cX_{\hf}$,  viewing the resulting object as a ``square root" of the $p$-adic $L$-function interpolating the central critical values  $L(f_x^\circ,V,k/2)$. 
 
% \begin{theorem}[Blanco-Chac\'on--Fornea]
 %\label{thm:padicL}
 %There is an element
 %$\Lp(\hf,V)\in {\rm Frac}(\Lambda_f)$ which is regular at $x_0\in\cX_{\hf}$ and satisfies 
 %$$
 %\Lp(\hf, V)(x) =  \mathcal{E}(f_x,V)  \cdot 
 %\mathcal{E}_0(f_x)^{-1} \cdot I(f_x^\circ,G),
 %$$
%for all  classical points
%$x\in \cX_f$  of weight $k =2t+2> 2$, where 
%\begin{equation}\label{Euler-Asai}
%\mathcal{E}(f_x,V) =  \left(1 -  \frac{p^{t}\alpha_G}{\alpha_x }\right) \left(1 -  \frac{p^{t}\beta_G}{\alpha_{x} }\right), 
%\qquad
%\cE_0(f_x) =  \left(1 - \frac{p^{k-1}}{\alpha^2_{x}}\right).
%\end{equation}  
%\end{theorem} 
%\begin{proof}[Sketch of proof]
%Let $d_1 := q_1 \frac{d}{dq_1}$ be the partial $d$ operator   which maps Hilbert modular forms of weight $(k_1,k_2)$ to nearly overconvergent Hilbert modular forms of weight $(k_1+2,k_2)$. The generating
%series
%\begin{equation}
%\label{eqn:theta-G} 
%d_1^t G_\fM(q_1,q_2) := \sum_{\nu\in (\fM\mathfrak{d}^{-1})_+} 
%\nu_1^t G((\nu)\mathfrak{d})   q_1^{\nu_1} q_2^{\nu_2} \ 
%\in \ M_{1+2t,1}^{\rm no}(\Gamma_0(\fD,\fM))
%\end{equation}
%is the $q$-expansion of a {\em nearly overconvergent} $p$-adic Hilbert modular form of weight $(1+2t,1)$.   
%Its diagonal restriction  
% $$ H_k^{\rm no} := d^t G_\fM(q,q) \in M_k^{\rm no}(\Gamma_0(DM))$$ 
% is a nearly overconvergent modular form of weight $k$ and tame level $DM$. Let 
% $$ H_k^\flat :=  e_{\rm ord}  {\rm Tr}^{DM}_M H_k^{\rm no} \in M_k(\Gamma_0(N))$$
%be the  classical modular form of weight $k$ and level $N$ obtained by applying to $H_k^{\rm no}$:
%\begin{enumerate}
%\item
%the trace map ${\rm Tr}^{DM}_M$ from overconvergent forms of  level $DM$ to level $M$;
%\item the ordinary projection $e_{\rm ord}$ from overconvergent modular forms of tame level $M$ to classical ordinary modular forms of level $N$.
%\end{enumerate} 
%The geometric principles evoked in the proof of
%\cite[Prop.~2.8]{DR1}
%imply that
%$$ H_k^\flat = e_{\rm ord}H_k,$$
%and hence $H_k^\flat$ 
%can be envisaged as a $p$-adic avatar of the modular form $H_k$
% of \eqref{eqn:def-Hk}. 
%Replacing $H_k$ by $H_k^\flat$ in
%\eqref{eqn:triple-period-C}
%leads to  an alternate  expression for $I(f_x^\circ,G)$ in terms of the period $\langle H_k^\flat, f_x\rangle  \langle f_x,f_x\rangle^{-1}$.
 % Although more genuinely $p$-adic in nature, 
%this quantity does not interpolate  $p$-adically to an analytic or even continuous function of $x\in \cX_{\hf}$, because the quantity $
 %\nu^t$ arising in the fourier coefficients of $d^t G_{\fM}$ are only analytic functions of $t$ when 
% $p\nmid \nu$.
% This motivates replacing the form
% $d_1^t G_\fM(q_1,q_2)$ by its
% $p$-depletion in 
% \eqref{eqn:theta-G}: \begin{equation}
%\label{eqn:theta-G-bis} 
%d_1^t G_\fM^{[p]}(q_1,q_2) := \sum_{\substack{\nu\in (\fM\mathfrak{d}^{-1})_+\\ p\nmid \nu}} 
%\nu_1^t G((\nu)\mathfrak{d})   q_1^{\nu_1} q_2^{\nu_2},
%\end{equation}
%a modular generating series whose fourier coefficients interpolate to  $p$-adic analytic functions of $t\in (\Z/(p^2-1)\Z) \times\Z_p$.
%Let 
%$$H_k^{[p]}  := e_{\rm ord}  {\rm Tr}^{DM}_M \left(d_1^t G_\fM^{[p]}(q,q) \right)  \in M_k(\Gamma_0(N))$$ 
%be the 
% ordinary projection of the trace to level $M$ of the diagonal restriction of this nearly overconvergent Hilbert modular form,
 % and set
% \begin{equation}
% \label{Lp-def} 
% \Lp(\hf,V)(x) := 
% \frac{\langle H_k^{[p]}, f_x\rangle}{\langle f_x,f_x\rangle}. 
% \end{equation}
% This quantity interpolates to a $p$-adic meromorphic function of $x$, denoted $\Lp(\hf,V)$.
%After specialising \cite[Theorem 1.3]{BF} 
 %to the ``weight $(k,1,1)$ line", we obtain
% $$ \Lp(\hf,V)(x) = 
% \cE(f_x,V) \cdot \cE_0(f_x)^{-1} \cdot \frac{\langle H_k, f_x^\circ\rangle}{\langle f_x^\circ, f_x^\circ\rangle},$$
% and Theorem \ref{thm:padicL} follows.
% \end{proof}
 
  \begin{remark}
It is instructive to compare the ``$p$-adic multiplier"
$\cE(f_x,V)\cE_0(f_x)^{-1}$ of Theorem  \ref{thm:padicL} 
 with the ostensibly more complicated
expression  in
\cite[Theorem 4.7]{DR1}, where $G$ is replaced by a pair $(g,h)$ of  elliptic modular forms of weight one, and 
    $V$ is replaced by the tensor product $V_{gh}$ of the   odd two-dimensional Artin representations attached to $g$ and $h$. In that setting, the factor that enters into the $p$-adic interpolation is given by
 \begin{equation}
 \label{eqn:dr-multiplier}  \cE_+(f_x,V_{gh}) \cdot \cE_0(f_x)^{-1} \cdot \cE_1(f_x)^{-1},
 \end{equation}
 where
\begin{equation}
\label{eqn:multiplier-artin}
 \cE_+(f_x,V_{gh}) =
 \prod_{i=1}^4 \left(1- \frac{p^t\gamma_i}{\alpha_x}\right),  
 \end{equation}
 with  $\gamma_1, \gamma_2,\gamma_3$ and $\gamma_4$    the eigenvalues  of $\sigma_p$ acting on $V_{gh}$, and
$$
 \cE_0(f_x) = \left(1-\frac{p^{k-1}}{\alpha_x^2}\right), 
 \qquad 
 \cE_1(f_x) = \left(1-\frac{p^{k-2}}{\alpha_x^2}\right).
$$
Replacing $V_{gh}$ by $V$ in 
\eqref{eqn:multiplier-artin}  and noting that $1$ and $-1$ occur among the   $\sigma_p$-eigenvalues on $V$,  we find
 \begin{equation}\label{factor-Euler}
 \frac{\cE_+(f_x, V)}{\cE_0(f_x) \cE_1(f_x)} =
 \frac{(1-\alpha_G p^t)(1-\beta_G p^t) \cE_1(f_x)}{\cE_0(f_x) \cE_1(f_x)} = \frac{\cE(f_x,V)}{\cE_0(f_x)},
 \end{equation}
 and hence the $p$-adic multipliers in \cite{BF} and
 \cite{DR1} are consistent.
 \end{remark}
 
% \begin{remark}
 %\label{rem:multiplier}
 %The definition of the multiplier $\cE_+(f_x, V_{gh})$ in
 %\eqref{eqn:multiplier-artin}
 %has an obvious extension where $V_{gh}$ is replaced by an arbitrary Artin representation, satisfying the following multiplicativity property:
 %\begin{equation}
 %\cE_+(f_x, V_1 \oplus V_2) = \cE_+(f_x,V_1) \cdot \cE_+(f_x,V_2).
 %\end{equation}
 %\end{remark}
 
The next result can be viewed as a Gross-Zagier formula for $\Lp(\hf,V)$ since it expresses its first derivative at $x=x_0$ in terms of the  $p$-adic logarithm of the  point $P_{V,f}$  of Conjecture \ref{main-conj}.
The weight map $\w$ is \'etale at $x_0$ and the standard local parameter 
at $k=2$ in $\cX$ induces a local parameter on $\cX_f$ at $x_0$, with respect to which the derivatives evoked in the following theorem are to be taken.

\begin{theorem}
\label{thm:gross-zagier}
We have
\begin{equation}\label{Invariant-ThetaDerivative}
 \frac{d}{d x}
   \Lp(\hf, V)_{|x=x_0} = C \lambda_{V,f}
\end{equation}
for some constant $C\in \kappa^\times$.
\end{theorem}

\begin{proof} 
The generating series
$G_\fM'(q_1,q_2)$
of 
\eqref{eqn:deformation-series} can be written as
$$ G_{\fM}' = \frac{d}{dt} \left(d_1^t G_{\fM}^{[p]}\right)_{t=0}.$$
The reader is cautioned that the letter $d$ appears in this equation in two different guises, whose distinct meanings should nonetheless be apparent from the context.
It follows that the  ordinary modular form $\Phi_V$ of
\eqref{eqn:def-Phi-N}
is given  by
\begin{equation}
\label{eqn:PhiNderiv} \Phi_V= \frac{d}{dk}(H_k^{[p]})_{k=2}.
\end{equation}
The modular forms $H_k^{[p]}$ are the classical specialisations of a $\Lambda$-adic family of ordinary modular forms. By
\eqref{Lp-def}, 
 the component of this $\Lambda$-adic form along the eigenform $\hf$   is equal to $\Lp(\hf,V) \cdot \hf$.
Since $\Lp(\hf,V)(x_0)=0$, 
projecting the 
 equality \eqref{eqn:PhiNderiv} to the $f$-isotypic component we obtain
 $$ \frac{d}{dx} \Lp(\hf,V)_{x=x_0} = \lambda_{V,f}$$
 up to a constant in $\kappa^\times$, as claimed.
 \end{proof}
 
 
 In view of \eqref{eqn:decomp-PhiN} we may thus recast Conjecture \ref{conj:dlr} in the following equivalent way.
 

\begin{conjecture}
\label{conj:dlr2}
The derivative of the twisted triple-product $p$-adic $L$-function at $x=x_0$ is given by
\begin{equation}
%% \label{eqn:decomp-PhiN}
  \frac{d}{dx} \Lp(\hf,V)_{|x=x_0} = \log_\omega(P_{V,f})\end{equation} 
% where $P_{V,f}\in J_0(N)(H)_f^V$. 
up to a multiplicative constant in $\kappa$.
   \end{conjecture}


   

   



%\begin{theorem}\label{vanish} [Not sure anymore about it] Assume $a = \alpha_G$ (in which case we have $a = \alpha_G = \beta_G$). Then\footnote{for every choice of $\eta$ %associated to a narrow ideal class $\lambda \in \Cl^+(F)$ and any choice of test vectors $ \testf, \, \hat{\gamma} = \hat\gamma_{G,\lambda} \in S_2(pM_\lambda N)$}
%$$
%\frac{\langle f, e_{\ord}(\gamma_G) \rangle }{\langle f, f \rangle} = 0.
%$$
%\end{theorem}

%\begin{proof}
%Under the running hypotheses of the previous section it follows that
% \begin{equation}\label{Invariant-ThetaDerivative}
% \frac{\partial}{\partial x}  \Lp^f(\mathbf{f}, \As(G))_{|x=x_0} = \frac{\langle f, e_{\ord}(\gamma_G)\rangle}{\langle f, f\rangle}.
% \end{equation}
 
% The Euler factor $\mathcal{E}_x  = \alpha^{-2}_{\mathbf{f}_x} (\alpha_{\mathbf{f}_x} - \alpha_G) (\alpha_{\mathbf{f}_x} - \beta_G)$  is a rigid-analytic function [Not sure anymore about %this latter claim]. Since it vanishes identically whenever $a = \alpha_G$ or $\beta_G$, the same holds for the $p$-adic $L$-function $ \Lp^f(\mathbf{f}, \As(G))$  in light of Theorem %\ref{padicL}. The statement thus follows from \eqref{Invariant-ThetaDerivative}.
%  \end{proof}
 
 
 \medskip
 
 \section{Hilbert theta series}\label{sec:CM}

Let $K/F$ be a quadratic extension of the real quadratic field $F$, and let 
$$\psi: G_K \lra \ \kappa^\times$$ be a finite order character of $K$. 
The case where $G = \theta_\psi$ is the Hilbert theta series over $F$ attached to $\psi$ provides a particularly enticing setting for Conjecture \ref{main-conj}, since the associated Asai representation then factors through a finite 
abelian extension  of (the Galois closure of) $K$.
The possibility of varying $\psi$ suggests  the  construction of a systematic collection of global points on elliptic curves defined over such abelian extensions.

To remain within the scope of the main conjecture in the introduction, it will be assumed throughout that
\begin{enumerate}
\item   $\psi$ is of mixed signature at any pair of real places of $K$ that lie over a common real place of $F$.  This ensures that
the Artin representation
$V_0$ is odd at both real places of $F$, and that
$G=\theta_\psi$ is a holomorphic Hilbert modular form of parallel weight one.
\item
The induced representation $V_0$ of $G_F$ satisfies the  assumption in equation
\eqref{eqn:transfer1} of the introduction, implying that $V$ is self-dual.
\end{enumerate}
We begin by spelling out what this  second condition on the self-duality of $V$  implies about $\psi$.  Let $\chi_{_{K/F}}$ be the quadratic character of $G_F$ attached to $K$.
\begin{lemma}
\label{lemma:self-dual-theta}
Let $V_0$ be the induced representation of $G_F$ attached to $\psi$, and let 
 $V$ denote its tensor induction to $\Q$.  
 If
 $$ {\rm Ver}_F^\Q (\chi_{_{K/F}}) \cdot {\rm Ver}_K^\Q(\psi) =1
 $$
 then $V$ is self-dual.
 \end{lemma}
 \begin{proof}
Since
 $$\det(V_0) = \chi_{_{K/F} }\cdot {\rm Ver}_K^F(\psi),$$
 it follows from the transitivity of the transfer map that
 $$ {\rm Ver}_F^\Q\det(V_0) = {\rm Ver}_F^\Q (\chi_{_{K/F}}) \cdot {\rm Ver}_K^\Q(\psi).
 $$
  Lemma \ref{lemma:self-dual-theta}
 now follows from \eqref{eqn:transfer1}.
 \end{proof}
The quadratic character
${\rm Ver}_F^\Q (\chi_{_{K/F}})$ of $G_\Q$ that arises in 
Lemma
 \ref{lemma:self-dual-theta}
can be described explicitly, according to the following analysis in three cases.

\medskip
\noindent
{\bf Case 1}. If $K/\Q$ is a biquadratic field, the transfer
map from ${\rm Gal}(K/\Q)$ to ${\rm Gal}(K/F)$ is the trivial homomorphism, and hence
\begin{equation}
\label{eqn:ver1}
{\rm Ver}_F^\Q (\chi_{_{K/F}}) = 1.
\end{equation}

\medskip
\noindent
{\bf Case 2}. If $K/\Q$ is a cyclic quartic extension, the transfer
map from ${\rm Gal}(K/\Q)$ to ${\rm Gal}(K/F)$
has kernel equal to ${\rm Gal}(K/F)$ and  
 induces an isomorphism from ${\rm Gal}(F/\Q)$ to ${\rm Gal}(K/F)$ by passing to the quotient.
Therefore
\begin{equation}
\label{eqn:ver2}
{\rm Ver}_F^\Q (\chi_{_{K/F}}) = \chi_{_{F/\Q}}.
\end{equation}

\noindent
{\bf Case 3}. If $K/\Q$ is a non-Galois quartic extension, then
 the normal closure $\tilde K$ of $K$ 
has Galois group isomorphic to the dihedral group $D_8$ of  order $8$.
It is a biquadratic extension of $F$, and hence $F={\tilde K}^\Pi$ where $\Pi\simeq \Z/2\Z\times \Z/2\Z$ is a Klein four-group in $D_8$. There are exactly  two other  subgroups of order $4$ in $D_8$: the cyclic group of order $4$  and a second Klein four-group, denoted $\Pi_\star$. Let $F_\star = \tilde K^{\Pi_\star}$ be the quadratic extension of $\Q$ associated to $\Pi^\star$ under the Galois correspondence.

A direct group-theoretic calculation reveals that
the transfer map $\phi_\Pi: D_8\lra \Pi$ has kernel $\Pi_\star$ and  identifies $D_8/\Pi_\star$ with the center $\{\pm 1\}$ of $D_8$ (viewed as a subgroup of $\Pi$), i.e., 
$$ \phi_\Pi(g) = \left\{
\begin{array}{rl}
1  & \mbox{ if } g\in \Pi_\star, \\
-1 & \mbox{ if } g\notin \Pi_\star.
\end{array} \right.
$$
It follows that
\begin{equation}
\label{eqn:ver3}
{\rm Ver}_F^\Q (\chi_{_{K/F}}) = \chi_{_{F_\star/\Q}}.
\end{equation}

\begin{corollary}
\label{cor:self-duality-theta}
The Asai representation
arising from the two-dimensional representation $V_0= {\rm Ind}_K^F\psi$ is self-dual 
if and only if
$$
\begin{cases} 
\Ver_K^{\Q}(\psi)=1, & \mbox{ when } K/\Q \mbox{ is bi-quadratic}, \\
 \Ver_K^{\Q}(\psi) = \chi_{_{F/\Q}}, & \mbox{ when } K/\Q \mbox{ is cyclic}, \\
 \Ver_K^{\Q}(\psi)  = \chi_{_{F_{\star}/\Q}}, & \mbox{ when } K/\Q \mbox{ is not normal}.
\end{cases}
$$
\end{corollary}
\begin{proof}
This follows by combining 
\eqref{eqn:ver1},
\eqref{eqn:ver2}, and
\eqref{eqn:ver3}, with
Lemma
\ref{lemma:self-dual-theta}.
\end{proof}
 
 \medskip

\subsection{Biquadratic extensions}

We now  focus further on the scenario where $K$ is a quadratic extension of $F$ that is  biquadratic over $\Q$. It then contains two further quadratic extensions of $\Q$,
denoted $K_1$ and 
$K_2 $. Because $F$ is a real quadratic field, the fields $K_1$ and $K_2$ are either both real, or both imaginary, depending on whether $K$ is a totally real or CM extension of $F$.

Let 
$$\psi_1=\mathrm{Ver}_K^{K_1}(\psi), \qquad \psi_2=\mathrm{Ver}_K^{K_2}(\psi)$$ denote the transfers of $\psi$ to $G_{K_1}$ and $G_{K_2}$ respectively. 
\begin{lemma}
The characters $\psi_1$ and $\psi_2$ are ring class characters of  their respective quadratic fields. If $K_1$ and $K_2$ are  real, then
 $\psi_1$ and $\psi_2$ have opposite parity, i.e.,  $\psi_2$ is totally odd when $\psi_1$ is totally even, and vice-versa.
 \end{lemma}
 \begin{proof}
Corollary
\ref{cor:self-duality-theta}
 implies that  
 ${\rm Ver}_{K_1}^\Q(\psi_1) = {\rm Ver}_{K_2}^\Q(\psi_2) = 1$, 
 and
 this implies the first assertion. The second assertion follows from the
 assumption that $\psi$ has mixed signature at any pair of  real places of $K$ that lie above a common real place of $F$.
\end{proof}

Recall that $V_0:= {\rm Ind}_K^F(\psi)$ is the representation of $G_F$ induced from $\psi$, and that $V= {\rm Ind}^{\otimes}(V_0)$ is
the associated Asai representation of 
$G_\Q$.
\begin{proposition}\label{decomp}
The  Artin representation 
$V$ decomposes as
$$
V = V_1 \oplus V_2, \qquad \mbox{ where } \quad
V_1 := \mathrm{Ind}_{K_1}^{\Q}(\psi_1), \quad  V_2 :=  \mathrm{Ind}_{K_2}^{\Q}(\psi_2).
$$
\end{proposition}
\begin{proof}
Let   $\sigma$, $\sigma_1$,   $\sigma_2$, $\tau$, $\tau_1$ and $\tau_2$  denote the non-trivial elements of
${\rm Gal}(K/F)$,  ${\rm Gal}(K/K_1)$,  ${\rm Gal}(K/K_2)$,
${\rm Gal}(F/\Q)$, ${\rm Gal}(K_1/\Q)$,  and
${\rm Gal}(K_2/\Q)$  respectively, as summarized in the following field diagram:
\begin{equation}
\label{eqn:field-diagram}
\xymatrix{
   & K \ar@{-}[dl]_{\sigma_1}  \ar@{-}[d]^{\sigma} \ar@{-}[dr]^{\sigma_2 } &  
     \\
  K_1 \ar@{-}[dr]_{\tau_1} & F \ar@{-}[d]^\tau  & K_2 \ar@{-}[dl]^{\tau_2} \\ 
  & \Q & }
   \end{equation}
 The character of the induced representation $V_0$ is given by the formula
$$ \chi_{_{V_0}}(g) = \left\{
\begin{array}{cl}
\psi(g) + \psi^{\sigma}(g) & \mbox{ if } g\in G_K;\\
0 & \mbox{ if } g\in G_F-G_K,
\end{array}\right.
$$
and the character of 
$V$  is given by
$$ \chi_{_{V}}(g) = \left\{
\begin{array}{cl}
\chi_{_{V_0}}(g) \chi_{_{V_0}}(g^\tau)    & \mbox{ if } g\in G_F;\\
\chi_{_{V_0}}(g^2) & \mbox{ if } g\in G_\Q-G_F.
\end{array}\right.
$$
These two formulae can be used to show that
\begin{equation}
\label{eqn:char-ident}
 \chi_{_V}(g) = \chi_{_{V_1}}(g) + \chi_{_{V_2}}(g),
 \end{equation}
  according to the following division into three cases:

\medskip
\noindent
{\bf Case 1}: $g\in G_K$.
Then
\begin{eqnarray*}
 \chi_{_V}(g) &=& \chi_{_{V_0}}(g) \chi_{_{V_0}}(g^\tau) \\
 &=&  
(\psi(g) + \psi^\sigma(g))(\psi^{\sigma_1}(g)+\psi^{\sigma_2}(g)) \\
&= &
(\psi\psi^{\sigma_1}  + \psi^\sigma\psi^{\sigma_2})(g) + (\psi \psi^{\sigma_2} + \psi^\sigma\psi^{\sigma_1})(g)  \\
&=& (\psi_1 + \psi_1^{\tau_1})(g) + (\psi_2 + \psi_2^{\tau_2})(g) \\
&=& \chi_{_{V_1}}(g) + \chi_{_{V_2}}(g).
\end{eqnarray*}

\smallskip
\noindent
{\bf Case 2}: $g\in G_F - G_K = (G_\Q-G_{K_1})\cap (G_\Q-G_{K_2})$. Then
$$ \chi_{_V}(g) = \chi_{_{V_0}}(g) \chi_{_{V_0}}(g^\tau) = 0.$$
Since $g$ belongs to neither $G_{K_1}$ or $G_{K_2}$,  we also have
$$  \chi_{_{V_1}}(g)  =   \chi_{_{V_2}}(g)=0,$$
and \eqref{eqn:char-ident}
follows.

\smallskip
\noindent
{\bf Case 3}. $g\in G_\Q -G_F = (G_{K_1}-G_{K_2}) \sqcup (G_{K_2}-G_{K_1})$. Then $g^2$ belongs to $G_K$ and hence
\begin{eqnarray*}
\chi_{_V}(g) &=& \chi_{_{V_0}}(g^2) = \psi(g^2) + \psi^\sigma(g^2)  \\
&=& \left\{
\begin{array}{cl}
 (\psi\psi^{\sigma_1}+ \psi^\sigma \psi^{\sigma_2})(g)  = (\psi_1+\psi_1^{\tau_1})(g)& \mbox{ if } g\in G_{K_1}- G_{K_2} \\
  (\psi\psi^{\sigma_2}+ \psi^\sigma \psi^{\sigma_1})(g)  = (\psi_2 + \psi_2^{\tau_2})(g) & \mbox{ if } g\in G_{K_2}- G_{K_1} 
  \end{array}\right.  \\
  &=& \left\{
\begin{array}{cl}
 \chi_{_{V_1}}(g)  + 0 & \mbox{ if } g\in G_{K_1}- G_{K_2} \\
  0 + \chi_{_{V_2}}(g) & \mbox{ if } g\in G_{K_2}- G_{K_1} 
  \end{array}\right.  \\
  &=& \chi_{_{V_1}}(g) + \chi_{_{V_2}}(g).
  \end{eqnarray*}
 The proposition follows.
  \end{proof}


As in Section \ref{sec1}, let $E$ be an elliptic curve over $\Q$ of conductor $N=pM$ where $p$ is a prime that remains inert in $F$ and $M\geq 1$ is a positive number all whose prime divisors split in $F$. Letting $D_{\psi_1}$, $D_{\psi_2}$ denote the conductors of the central characters of $\psi_1$, $\psi_2$, we further assume that $(N, D_F D_{\psi_1} D_{\psi_2})=1$. Finally, let us also assume throughout this section that hypotheses $(H_p)$ and $(H_M)$ in Theorem \ref{thm:padicL} are in place for the eigenform $f$ associated to $E$.

 
When combined with  the Artin formalism for  Hasse-Weil-Artin $L$-series,
Proposition \ref{decomp}
implies 
 the
factorisation
\begin{eqnarray}
\label{fac}
L(f_x^\circ, V,k/2) &=& 
L(f_x^\circ, V_1,k/2) \cdot
L(f_x^\circ, V_2,k/2) 
\\ &=&
\nonumber
L(f_x^\circ/K_1, \psi_1, k/2) \cdot  L(f_x^\circ/K_2, \psi_2, k/2)
\end{eqnarray}
for all classical $x\in \cX_{\hf}$,
and
suggests that a similar principle might  apply to   $p$-adic $L$-series  interpolating these special values.

In order to state this more precisely, let $\Omega_x^+$ and $\Omega_x^-$ be the real and imaginary 
periods attached to $f_x^\circ$, as introduced e.g. in \cite[p.\,488]{Hida}.
As explained in loc.\,cit., \cite[\S 1.4]{Ish} and \cite[\S 1.1]{BD2}, these periods can be chosen in such a way that 
\begin{equation}\label{ProductPeriods}
\Omega_x^{+} \Omega^{-}_{x} =  \langle f_x^\circ, f_x^\circ\rangle.
\end{equation}

Let $D_1$ and $D_2$ denote the discriminants of $K_1$ and $K_2$ respectively.
Following  \cite[(93)]{BD2} and \cite[Theorem 3.5]{BD3}, 
the algebraic parts of the special values attached to $f_x^\circ$ and the ring class characters $\psi_i$  of  $K_i$ ($i=1,2$)  are obtained  by setting
\begin{equation}\label{Lalg-BD}
L^{\mathrm{alg}}(f^\circ_x/K_i,\psi_i,k/2) = \frac{(\frac{k-2}{2}!)^2 D_i^{\frac{k-1}{2}}}{(2\pi)^{k}\,\Omega_{x,\psi_i}} L(f^\circ_x/K_i,\psi_i,k/2),
\end{equation}
where the period $\Omega_{x,\psi_i}$ is given as in \cite[Definition 3.4]{BD3} by
 $$\Omega_{x,\psi_i} = 
 \left\{\begin{array}{cl}
 \Omega_x^+ \Omega_x^-
 & \mbox{ if } D_i<0; \\
 (\Omega_x^+)^2 
 & \mbox{ if } D_i>0 \mbox{ and } \psi_i 
 \mbox{ is even}; \\
 (\Omega_x^-)^2 
 & \mbox{ if } D_i>0 \mbox{ and } \psi_i \mbox{ is odd}. 
 \end{array}\right.
 $$
 
Note that in our statement of \eqref{Lalg-BD} we have formulated a different power of $\pi$ with respect to \cite{BD2} and \cite{BD3}, which is due to the different normalizations in the definition of the Petersson scalar product adopted in loc.\,cit.\,versus \cite{Ish}.
 
Comparing these definitions with the one in 
Theorem~\ref{thm:padicL}, we see that for all classical $x\in \cX_{\hf}$ of
weight $k>2$:
\begin{equation}\label{factor-alg}
L_{\rm alg}(f_x^\circ, V, k/2) \sim
L_{\rm alg}(f_x^\circ/K_1,\psi_1,k/2)  \cdot 
L_{\rm alg}(f_x^\circ/K_2,
\psi_2,k/2).
\end{equation}

     
   The generalised Heegner hypothesis
   \eqref{eqn:heegner-hyp}
     for $V$ 
   implies that all the primes $\ell|M$ have the same splitting behaviours in $K_1$ and $K_2$, since they are split in $F/\Q$.
 Let
 $$ \Sigma := \{ \ell | M\infty \mbox{ such that } 
 \varepsilon_{K_i}(\ell)=-1 \}
 $$ 
 be the set of places of $\Q$ dividing $M\infty$ at which the quadratic field $K_1$ (and hence also $K_2$) is inert.
  
 The sign in the functional equations for $L(f_x^\circ,V_1,s)$ and $L(f_x^\circ, V_2,s)$ are
  controlled by the
 parity of the cardinality of $\Sigma$:
$$ {\rm sign}(f_x^\circ,V_1) = {\rm sign}(f_x^\circ, V_2) = 
(-1)^{\# \Sigma}.$$
When $\#\Sigma$ is odd, the central critical values $L(f_x^\circ/K_i,\psi_i,  k/2)$ vanish identically, and one has no  resort but to set
 \begin{equation}
 \label{eqn:vanish-fiat} \Lp(\hf,V_i) = 0.
 \end{equation}
When $\#\Sigma$ is even,
the special values $L(f_x^\circ,V_i,k/2)$  can be expressed in terms  of elementary quantities attached to optimal embeddings of  orders in $K_i$ (of  suitable 
conductor,   equal to the conductor of the character 
$\psi_i$)   in the quaternion algebra $B$ ramified at $\Sigma$. This explicit expression forms the basis for the construction of $p$-adic $L$-functions
$\Lp(\hf,V_i)$  ($i=1,2$) on $\cX_f$. 

This $p$-adic $L$-function
is constructed in 
\cite{BD2} and \cite{BD3} when $K_i$ is imaginary and real respectively, where they are
 also denoted $\Lp(\hf/K_i,\psi_i)$. They
 satisfy the interpolation property
\begin{equation}\label{intprop}
 \Lp(\hf,V_i)(x)^2 \sim \frac{\cE_+(f_x^\circ,V_i)^2}{\cE_0(f_x^\circ) \cE_1(f_x^\circ)} 
L_{\rm alg}(f_x^\circ, V_i,k/2), \qquad i=1,2,
\end{equation}
where $\cE_+(f_x^\circ,V_i)$ is the 
$p$-adic multiplier attached to $f_x^\circ$ and $V_i$, as defined in \eqref{eqn:multiplier-artin} after replacing $V_{gh}$ with $V_i$. Note that
\begin{equation}\label{Artin-Euler}
 \cE_+(f_x, V_1 \oplus V_2) = \cE_+(f_x,V_1) \cdot \cE_+(f_x,V_2).
 \end{equation}
 

For imaginary quadratic fields, the construction is detailed in 
 \cite[\S 3.2]{BD2}, and generalised in \cite[\S 3]{Mok}. For real quadratic fields, see \cite[\S 3.2]{BD3} and  its further refinements given  in \cite[\S 4.4, 4.5]{LV}.  We also refer to the work of Hern\'andez and Molina \cite{HM} for a recent construction of these  $p$-adic $L$-functions that applies to the    general setting considered here in all cases.



  
\begin{theorem}\label{thm:factorisation-padicL} 
 There is a factorization of $p$-adic $L$-functions
 $$
\Lp(\hf, V) \sim  \Lp(\hf/K_1, \psi_1) \cdot  \Lp(\hf/K_2, \psi_2),
 $$
 where $\sim$  is defined in \eqref{eqn:sim}. \end{theorem}
 
 \begin{proof}
 Note first that Theorem \ref{thm:padicL} is in force thanks to our running hypotheses. If $\#\Sigma$ is odd, the right-hand side is identically $0$, and so is the left-hand side in light of Theorem \ref{thm:padicL} and \eqref{fac}. When $\#\Sigma$ is even, it follows from Theorem \ref{thm:padicL} and equations \eqref{factor-Euler}, \eqref{factor-alg}, \eqref{intprop} and \eqref{Artin-Euler}  that both sides of the factorization claimed in the statement are rigid-analytic functions taking the same values in a Zariski dense subset of $\cX_f$, up to an admissible function. This yields the theorem.
 \end{proof}

  Since the prime $p||N$ is inert in $F/\Q$ and $\Gal(K/\Q)=\Z/2\Z \times \Z/2\Z$, it follows that $p\cO_F$  necessarily
   splits in $K/F$. In particular, $p$   has different splitting behaviours in the quadratic fields $K_1$ and $K_2$. From now on in this section, these two fields will be ordered in such a way that
   $p$ remains inert in $K_1$ and splits in $K_2$. (Note that this convention is not followed in Section \ref{Sec-Experiments}.)
   This implies that $\sigma_p$ acts on $V_1$ with eigenvalues $\pm 1$, and  on $V_2$ with eigenvalues $\alpha_G$ and $\beta_G = \alpha_G^{-1}$, and that
  \begin{equation}
  \label{eqn:sign-change}
   {\rm sign}(E,V_1) = -{\rm sign}(f_x^\circ,V_1),
   \qquad {\rm sign}(E,V_2) = {\rm sign}(f_x^\circ,V_2)
   \end{equation}
  for all classical $x\in \cX_{\hf}$ of
weight $k>2$.
  
  
The following theorem is
the main result of this section:
\begin{theorem}\label{conj-CM} % AL to VR: the label of this theorem is a little inaccurate, but if you change the label note I do refer to it a lot in the examples section
Assume that either
\begin{enumerate}
\item  $K_1$ and $K_2$ are imaginary quadratic fields, or
\item  $K_1$ and $K_2$ are real quadratic fields, and the conjectures of  \cite[\S 5]{Dar}  on Stark-Heegner points 
hold for $(K_1, \psi_1)$. 
\end{enumerate}
Then Conjecture \ref{main-conj} is true. 
\end{theorem}
 \begin{proof}
  By Theorem 
  \ref{thm:gross-zagier}, the scalar $\lambda_{V,f}$ is equal to the first derivative of  
  $\Lp(\hf,V)$ at $x=x_0$ up to a multiplicative constant in $\kappa^\times$. 
  
  Assume first $\#\Sigma$ is odd. As already argued above, in this case $\Lp(\hf,V)$ vanishes identically on $\cX_{\hf}$ and hence $\lambda_{V,f} = 0$. This is aligned with Conjecture \ref{main-conj}, as conditions (I) and (II) of that conjecture are not both fulfilled. Indeed, if condition (I) were true, namely $\ord_{s=1} L(E,V,s)=1$, it would follow from  \eqref{eqn:sign-change} that
 $$ {\rm sign}(E,V_1)=1,\qquad {\rm sign}(E,V_2) = -1,$$
 and therefore
 $$  L(E,V_1,s)\ne  0, \qquad
 \ord_{s=1} L(E,V_2,s) = 1, $$
 so that $W_0 \supseteq V_1$ and $W_1 \subseteq V_2$ in the notation of Conjecture \ref{main-conj}. But $\sigma_p$ acts with eigenvalues $1$ and $-1$ on $V_1$, and therefore condition (II) would not be met.



Assume now that $\#\Sigma$ is even. By Theorem \ref{thm:factorisation-padicL},   $\Lp(\hf,V)$ has a simple zero at $x=x_0$ if and only if 
 \begin{equation}
 \label{eqn:sumord}
 \ord_{x=x_0} \Lp(\hf,V_1) + \ord_{x=x_0} \Lp(\hf,V_2) =1.
 \end{equation} 
 Since 
 $$ {\rm sign}(E,V_1)=-1,\qquad {\rm sign}(E,V_2) = 1,$$
equation  \eqref{intprop} implies the $p$-adic $L$-function 
 $\Lp(\hf,V_1)$ vanishes at $x_0$ and hence
  \eqref{eqn:sumord} is equivalent to 
  \begin{equation}
 \label{eqn:sumord-bis}
 \ord_{x=x_0} \Lp(\hf,V_1) =1, \qquad   \Lp(\hf,V_2)(x_0) \ne 0.
 \end{equation} 


Note that $\cE_+(f_x^\circ,V_2) = (1-\frac{\alpha_G}{\alpha_x}) (1-\frac{\beta_G}{\alpha_x})$. Hence it follows again from \eqref{intprop} that $\Lp(\hf,V_2)(x_0) \ne 0$ if and only if 
\begin{equation}\label{nonvanV2}
L(E/K,\psi_2,1) \ne 0 \quad \mbox{ and } a_p \ne \alpha_G, \beta_G.
\end{equation}
%and recall 
%$$
%\frac{\cE_+(f_x^\circ,V_2)^2}{\cE_0(f_x^\circ) \cE_1(f_x^\circ)} = \frac{(1-\frac{\alpha_G}{\alpha_x})^2 (1-\frac{\beta_G}{\alpha_x})^2}{ (1-\frac{p}{\alpha^2_x}) (1-\frac{1}{\alpha^2_x})}
%$$
 
Besides, by  \cite[\S 4.4]{BD2}, \cite[\S 3]{Mok}, \cite[\S 4]{BD3}, \cite[Theorem 4.31]{LV}, \cite[Theorem 8.2]{HM},  we have
\begin{equation}\label{padicGZ}
\frac{d}{dx}\Lp(\hf,V_1)_{x=x_0} = C_1
\logtate(x_{E,V_1} + a_p x^{\sigma_p}_{E,V_1}), \quad C_1\in \kappa^\times
%\qquad 
%\Lp(\hf,V_2) = C_2, \quad C_1, C_2\in \kappa^\times
\end{equation}
where $x_{E,V_1}$ is a local point in $E(\Q_{p^2})$, with coordinates in the quadratic unramified extension $\Q_{p^2}$ of $\Q_p$. Note $x_{E,V_1} + a_p x^{\sigma_p}_{E,V_1}$  lies in the $a_p$-eigenspace for the action of $\sigma_p$. 

Moreover, $x_{E,V_1}$ is
\begin{itemize}
\item the image of a global point $Q_{E,V_1} \in E(H)^{V_1}$ under the natural embedding $E(H) \hookrightarrow E(\Q_{p^2})$, given by a Heegner point on a Shimura curve attached to the odd set 
$\Sigma\cup \{p\}$ of places, when $K_1$ is an imaginary quadratic field;
\item a Stark-Heegner point attached to the $p$-adic uniformisation of $E$ via rigid analytic cocycles, which is predicted to arise from a global point $Q_{E,V_1} \in E(H)^{V_1}$ similarly as above, by the conjectures of \cite[\S 5]{Dar}. 
% (Since the set $\Sigma\cup\{p\}$ does not contain $\infty$, the geometric object which would play the role of a Shimura curve when $\infty\in \Sigma$ is ostensibly unavailable, leading to a construction that is purely conjectural at present.)
\end{itemize}

The Gross-Zagier formula in the scenario where $K$ is imaginary (\cite{gross-zagier}) and its conjectural extension to real quadratic fields (cf.\,\cite[\S 5]{Dar}) imply that
$L(f,V_1,s)$ has a simple zero at $s=1$ if and only if $Q_{E,V_1}$ is not trivial in $E(H)\otimes \kappa$. 

If $V_1$ is irreducible (which is precisely the case when $\psi_1^2\ne 1$), then the image of $E(H)^{V_1}$ in $E(\Q_{p^2}) \otimes \kappa$ is a two-dimensional $\kappa$-vector space spanned by  $x_{E,V_1}$ and $x^{\sigma_p}_{E,V_1}$, and hence $Q_{E,V_1}$ is non-trivial if and only if $P_{E,V_1} := Q_{E,V_1} + a_p Q^{\sigma_p}_{E,V_1}$ is non-trivial, and this in turn is equivalent to the non-vanishing of \eqref{padicGZ}.  Note that the irreducible component $W_1$ of $V$ introduced in Conjecture \ref{main-conj} is $V_1$ in this case.

If $V_1$ decomposes as the sum of a pair of $1$-dimensional sub-representations $V_1 = W_1 \oplus W_1'$, then we may order them accordingly to the notations in Conjecture \ref{main-conj}, so that $\ord_{s=1} L(E,W_1,s)=1$ and $L(E,W_1',1) \ne 0$. 
%A generator of $E(H)^{W_1}$ is given by $R_{W_1} := Q_{E,V_1} + \epsilon Q_{E,V_1}^{\sigma_p}$, where $\epsilon = \pm 1$ stands for the eigenvalue of  $\sigma_p$ acting on $W_1$. 
Let $\epsilon = \pm 1$ denote the eigenvalue of  $\sigma_p$ acting on $W_1$. In the favorable case where $\epsilon = a_p$, then $P_{E,V_1} := Q_{E,V_1} + a_p Q_{E,V_1}^{\sigma_p}$ is a generator of $E(H)^{W_1}$, and its  image in $E(\Q_{p^2})$ is the local point appearing in the right-hand side of \eqref{padicGZ}. In particular  \eqref{padicGZ} again does not vanish. In the unfortunate case that $\epsilon = -a_p$,  the global point $Q_{E,V_1} + a_p Q_{E,V_1}^{\sigma_p}$ lies in $E(H)^{W'_1} = 0$ and hence \eqref{padicGZ} vanishes.


Summing up, we conclude that $\lambda_{V,f} \ne 0$
if and only if 
\begin{enumerate}
\item[(i)]
$L(E/K,\psi_2,1) \ne 0$ and $\ord_{s=1} L(E/K,\psi_1,s)= 1$,

\item[(ii)] $a_p \ne \alpha_G, \beta_G$, 

\item[(iii)] the eigenvector of eigenvalue $a_p$ lies in $W_1 \subseteq V_1$.
\end{enumerate}

% In the scenario \eqref{final}, it thus holds that $a_p$ occurs with multiplicity $1$ as eigenvalue of $\sigma_p$ acting on $V$, and its associated eigenvector lies in $W_1$. 

Since $L(E,V,s)= L(E/K,\psi_1,s) L(E/K,\psi_2,s)$, we have proved (subject to the conjectures in \cite[\S 5]{Dar} in the case of real quadratic fields) that $\lambda_{V,f} \ne 0$ if and only if conditions (I) in (II) in Conjecture \ref{main-conj} hold, and in that case $\lambda_{V,f} = \logtate(P_{V,f})$ for a point $P_{V,f}$ that forms a basis of the one-dimensional subspace
of the image of $E(H)^{W_1}$ on which $\sigma_p$ acts with eigenvalue $a_p$.
%In both cases,
% \eqref{eqn:sumord-bis} is therefore equivalent to the non-triviality of the point
%$P_{f,V_1}$, combined with the   fact that
%$$ \log_{E}(P_{f,V}) = \log_E(P_{f,V_1}) \pmod{\kappa^\times}.$$
\end{proof}
 
 
 
 
%\begin{remark}
%The scenario where $\#\Sigma$ is odd represents a prototypical instance where the point $P_{V,f}$  vanishes systematically  even  when 
% $ \ord_{s=1} L(E,V,s)=1$.
%For in this case,  
% ${\rm sign}(f_x^\circ,V_1)= {\rm sign}(f_x^\circ,V_2)=-1$, and the $p$-adic $L$-functions $\Lp(\hf,V_1)$ and $\Lp(\hf,V_2)$ vanish identically on $\cX_{\hf}$.
% The same is therefore true of $\Lp(\hf,V)$, by
% Theorem \ref{thm:factorisation-padicL}, and hence the local point $P_{V,f}$ of Theorem 
 %\ref{thm:gross-zagier}  is trivial. This is accounted for by the fact that  Condition (II) of Conjecture \ref{main-conj} fails to hold. Indeed, by   \eqref{eqn:sign-change}, we have
% $$ {\rm sign}(E,V_1)=1,\qquad {\rm sign}(E,V_2) = -1,$$
% and therefore
% $$  L(E,V_1,s)\ne  0, \qquad
% \ord_{s=1} L(E,V_2,s) = 1, $$
% so that $W_0=V_1$ and $W_1=V_2$ in the notation of Conjecture
% \ref{main-conj}.
% But $\sigma_p$ acts with eigenvalues $1$ and $-1$ on $V_1$, and therefore condition (II) is not met.
 %\end{remark}



%\subsection{Cyclic quartic fields}

%The setting where $K$ is a cyclic quartic field is of considerable interest, insofar as it ostensibly lies  outside the scope of 
% Theorem  xxxx and the machinery of
%Heegner and Stark-Heegner points.

%A numerical experiment testing the validity of 
%Conjecture \ref{main-conj}
%in this scenario is documented in section
%\ref{subsec:cyclic-quartic}.



%\subsection{Non Galois quartic fields}

%Even more intriguing perhaps is the case where $K$ is a non-Galois quartic extension of $\Q$. A few numerical experiments testing Conjecture \ref{main-conj} in this scenario are described in Section
%\ref{sec:non-galois}.

  
 
\medskip
\section{Numerical experiments}\label{Sec-Experiments}

In this section we present some numerical experiments which both illustrate various aspects of
Theorem \ref{conj-CM} 
 and give further evidence in support of the broader Conjecture \ref{main-conj}. We begin by considering
examples illuminating Theorem \ref{conj-CM} in the case where $K$ is a biquadratic field. These examples were computed prior
to the formulation of   Conjecture \ref{main-conj} and the proof of Theorem \ref{conj-CM}, and were key to suggesting their finer detail. They also suggest an analogous  but subtly different conjecture when the nearly overconvergent family interpolating 
$d^t G_{\fM}$ is replaced by a  Hida family  of parallel weight specialising to $G$ in weight $(1,1)$.

\subsection{Biquadratic extensions: the CM setting}
\label{sec:bi-CM}

\subsubsection{The basic set-up}
\label{sec-exs-dihedralCM-setting}

Let $D_1$ and $D_2$ be negative coprime fundamental discriminants and let $D_F := D_1 \cdot D_2 > 0$. Assume that these discriminants are odd  and that  $D_1, D_2 <-3$. Define as in 
\eqref{eqn:field-diagram}
\[ F:= \Q(\sqrt{D_F}),\quad K_1 :=\Q(\sqrt{D_1}),\quad
K_2 := \Q(\sqrt{D_2}), \quad \mbox{ and } \quad K := \Q(\sqrt{D_1},\sqrt{D_2}).\]
The biquadratic field $K$ is an odd {\em genus field} of the real quadratic field $F$:  an unramified CM quadratic extension  of  $F$  that  is also abelian over $\Q$, with Galois group isomorphic to $\Z/2\Z\times \Z/2\Z$.
 
We consider the simplest non-trivial setting, in which the  quadratic fields $K_1$, $K_2$ and $F$ have
class numbers $3$, $1$ and $1$  respectively.
The  narrow class number of $F$ is then equal to  $2$, and $K$ is the narrow Hilbert class field of $F$.
 The extension $K/F$  has (relative) class number $3$.

The Hilbert class field $H$ of the quartic field $K$ is a cyclic cubic extension of $K$, and a  degree $12$ Galois extension of $\Q$ which admits six
irreducible Artin representations: the four one-dimensional representations ${\bf 1}$,  $\varepsilon_{D_1}$, $\varepsilon_{D_2}$,
and $\varepsilon_{D_F}$,
attached to the quadratic Dirichlet characters of conductors $1$, $D_1$, $D_2$ and $D_F$ respectively,
 and two irreducible two-dimensional representations
 $$ W = {\rm Ind}_{K_1}^\Q \chi, \qquad W\otimes \varepsilon_{D_2}, $$
 where $\chi$ is any of the two  cubic unramified characters of $K_1$.  
   
   Let 
$$\psi : {\rm Cl}_{K/F} \rightarrow L^\times, \qquad L := \Q(\zeta_3) $$ 
be a non-trivial cubic character of the relative class group of $K/F$. The induced representation
$V_0 := {\rm Ind}_K^F(\psi)$ is a 
 two-dimensional representation of $G_F$, whose tensor induction to $\Q$ is given by
$$V := {\rm Ind}^\otimes(V_0) =  {\bf 1}  \oplus \varepsilon_{D_2} \oplus W. $$
 The   Artin representation 
$V_0$ induced by $\psi$ corresponds to a Hilbert theta series $G=G_\psi$  which is
described as a function on the ideals of $F$ by the formula
$$ G(I) = \sum_{\substack{J\in I(K),\\ {\rm Norm}(J) =I}}
\psi(J).$$
Recall that the construction of a weight $(1,1)$ modular generating  series $\theta_\psi$ attached to $G$ 
also depends on the choice of an ideal $\fM:=(\eta)$
in $O_F$. In the notation of the introduction,    the theta series $\theta_\psi=G_{\fM}(\tau_1,\tau_2)$   is given by the formula in  (\ref{eqn:modular-gen-series}).
Moreover, the $p$-depletion of $\theta_\psi$  is the weight $(1,1)$ specialisation of a natural
$p$-adic  {\it theta family}, given by $d_1^t G_\fM^{[p]}$ in  (\ref{eqn:theta-G-bis}). (Beware the two different uses of the term ``theta'' here.)
If $\fM=(\nu)$ belongs to the trivial narrow ideal class of $F$, i.e., if $\nu$ is totally positive,  then the  class of 
$(\nu)\fd$ in the   class group ${\rm Cl}_F$ is not a norm from ${\rm Cl}_K$ and hence
the sum defining $G((\nu)\fd)$ is empty, 
so that   the $p$-adic theta family $d^tG_{\fM}^{[p]}$ vanishes
identically.
So it is necessary to choose an ideal $(\eta)$ whose associated narrow ideal class  is non-trivial, i.e., for which
${\rm Norm}(\eta) < 0$. Note there is no element in $\cO_F$ of norm $-1$,
and hence that the integer $|{\rm Norm}(\eta)|$ is strictly greater than $1$. Since $K/F$ is unramified, the
diagonal restriction of the theta series $\theta_\psi$ lies in $S_2(|{\rm Norm}(\eta)|)$. In the notation of the
introduction, we have 
$$M = |{\rm Norm}(\eta)|, 
\qquad D = 1.$$
Note that all the primes dividing  $M$  must necessarily split in $F$.
  
Let $p$ be a prime which is inert in $F$. We consider elliptic curves $E$ of square-free conductor $pM$ with
associated eigenforms $f_E \in S_2(pM)$.  
 Since $pM$ satisfies 
 condition \eqref{eqn:heegner-hyp}
 of the introduction, 
  the Asai representation $V$ is expected to occur with odd multiplicity in the Mordell-Weil group
of $E$.

Let us assume now that this multiplicity is one. This implies that the rank patterns  
$$ (\ \ {\rm rank} (E(\Q)),   \ \  {\rm rank} (E^{D_2}(\Q)),  \ \ \dim \hom_{G_\Q} (W,E(H)_L) \ \ ) $$
are  equal to either $(1,0,0)$, $(0,1,0)$ or
$(0,0,1)$. The parities of ranks on each component which occur are correlated to the eigenvalues of $\sigma_p$ on the
Asai representation, and in the rank one setting the following three scenarios are the only ones that can occur:
\[
\begin{array}{l|l}
\mbox{Frobenius eigenvalues} & \mbox{Asai ranks} \\
\hline
(1,-1,(\zeta_3,\zeta_3^{-1})) & (1,0,0) \mbox{ or } (0,1,0)\\
(1,-1,(1,1)) & (1,0,0) \mbox{ or } (0,1,0) \\
(1,1,(1,-1)) & (0,0,1) 
\end{array}
\]
Write
\[ \lambda_{\theta} = \lambda_{\theta}(\psi,(\eta),p,f_E) \]
for the coefficient along $f_E$ of the diagonal restriction of the derivative ($G_\fM^\prime(\tau_1,\tau_2)$ in (\ref{eqn:deformation-series})) of the $p$-adic theta family through $\theta_\psi$.
Note that, in the notation of the introduction, the diagonal restriction of the derivative is given by $\Phi_{M}^{(p)}$ in (\ref{eqn:PhiMp}). Here
$D = 1$ and so it is not necessary to lower the level of $\Phi_{M}$ by taking its  trace  from level $DM$ to level $M$. (Note also that most of our examples are chosen so that a modification by a Hecke operator $T$ to annihilate forms of level $DM$ is not necessary, as the space is already zero, but see 
Example \ref{ex-cyclicquartic} for a situation where this is required.)

\begin{table}
\[
\begin{array}{l|l|c|c|c}
& \substack{\mbox{Frobenius}\\ \mbox{ eigenvalues}}& \mbox{Ranks} & a_p(E) & \substack{\mbox{Coefficient $\lambdatheta$} \\ \mbox{ along $f_E$.} } \\
\hline
1. & (1,-1,(\zeta_3,\zeta_3^{-1})) & (1,0,0)  & +1 &   \logtate(P^+)  \\
2. & (1,-1,(\zeta_3,\zeta_3^{-1})) & (1,0,0)  & -1 &   0  \\
3.& (1,-1,(\zeta_3,\zeta_3^{-1})) & (0,1,0)  & +1 &   0 \\
4.& (1,-1,(\zeta_3,\zeta_3^{-1})) & (0,1,0)  & -1 &   \logtate(P^-) \\
5.& (1,1,(1,-1))  & (0,0,1)  & +1 &   0 \\
6.& (1,1,(1,-1))  & (0,0,1)  & -1 &   \logtate(P_W^-) \\
7. & (1,-1,(1,1)) & (1,0,0) & \pm 1 &   0 \\
8. & (1,-1,(1,1)) & (0,1,0) & + 1 &   0 \\
9.& (1,-1,(1,1)) & (0,1,0) & - 1 &   \logtate(P^-)   
\end{array}
\]
\caption{Prediction for $\lambdatheta$ in the CM biquadratic setting.}\label{table:CMexs}
\end{table}


Conjecture \ref{main-conj} makes the predictions shown in Table \ref{table:CMexs}, which are proved in Theorem \ref{conj-CM}.  In this table, the global points 
$P^+$, $P^-$, and $P_W^-$  are  generators of $E(\Q)\otimes \Q$, 
of $E(K_2)\otimes \Q$, and
of the image of a generator of $E(H)_L^W$ on which $\sigma_p$ acts as $-1$.
 In the cases arising in the  fifth and sixth lines of the table, $E$ has rank $1$ over any cubic subfield of the Hilbert class field of $K_1$, and after 
choosing an embedding of one of these fields into $\Q_{p^2}$, there are global points $P_W^+$ and $P_W^-$ upon which 
$\sigma_p$ acts by $+1$ and $-1$, respectively.  In line 5, note  that $\lambda_\theta=0$  when $a_p(E)=1$,
 because  $+1$ does not have
multiplicity one in the Asai representation itself.


\subsubsection{Hida deformations of eigenforms} \label{sec-exs-dihedralCM-Hida}

We now take a small detour. This paper has as its focus {\it  theta deformations} of Hilbert eigenforms. But one can
also study the analogous question for Hida deformations of eigenforms. That is, the first infinitesimal deformation
of an eigenform in the $p$-adic Hida family in parallel weight through that eigenform (this family is unique, except in certain exceptional
cases). Studying this setting numerically and formulating a conjecture here was one of  the starting points for this paper.

The Hida setting has the practical disadvantage that currently one  only disposes of explicit Hida deformations
in the case of CM forms. The experimental evidence in this setting suggests though that letting 
\[ \lambdahida =   \lambdahida (\psi,(\eta),p,f_E) \]
denote the coefficient along $f_E$ of the diagonal restriction of the derivative of the $p$-adic Hida family in parallel weight through an ordinary stabilisation of $\theta_\psi$, 
the behaviour is exactly the same as for $\lambdatheta$ {\it except} one only need require
that $a_p(E)$ has multiplicity one in the component of the Asai representation which itself occurs with multiplicity
one in the Mordell-Weil group. For theta deformations, multiplicity one is required in the full Asai representation. As a consequence, 
if one replaces $\lambdatheta$ by $\lambdahida$ in Table 1  there is no change, {\it except}
in line 5   where  
$$\lambdahida = \logtate(P_W^+),$$
 and
in line 7 in which for rank $(1,0,0)$ and $a_p(E) = +1$ we get $\lambdahida = \logtate(P^+)$.


In the even more  special case of the dihedral quartic setting, there is also a {\it third} natural family to consider. 
To explain to the origin of this, note that the Hida family in parallel weight can be constructed explicitly by using grossencharacters
associated to $\psi$. The infinity type in this grossencharacter arises from a power of the norm map from $K$ down to the imaginary
quadratic field ($K_1$ or $K_2$) in which $p$ splits. Replacing this norm map by the one down to the imaginary quadratic field in
which $p$ is inert yields another natural and entirely explicit family. 

Curiously, in all examples that have been calculated, the   coefficient ``$\lambdanonhida$'' resulting from  this new family is $\logtate(P^{\pm})$ in the $\sigma_p$-eigenvalues $(1,-1,(\zeta_3,\zeta_3^{-1})) $ and ranks
$(1,0,0)$ and $(0,1,0)$ scenarios, regardless of the splitting type of $E$ at $p$. It would be interesting to 
formulate a systematic description of the behaviour of the
 coefficient ``$\lambdanonhida$''
%% in the rank $(0,0,1)$ setting.
in the remaining settings.

 
The same notations $\lambdatheta$ and (when computable) $\lambdahida$ shall be used beyond the CM biquadratic setting, in the examples in
Sections \ref{sec-exs-dihedralRM}, \ref{subsec:cyclic-quartic} and \ref{sec:non-galois}.

\subsubsection{Biquadratic extensions: numerical examples in the CM setting}\label{sec-exs-dihedralCM-exs}

We were able to numerically compute with all the class number $3$ and $1$ imaginary quadratic fields (satisfying our
mild hypothesis) and reasonably small values of $p$. We  focussed on
the relatively small primes  $p = 5,7,11$, which already gave rise  to a wealth of examples. The computations illustrated both the vanishing in rank $3$ settings, and
the proposed interpretation of $\lambdatheta, \lambdahida$ (and sometimes $\lambdanonhida$) in rank $1$ settings. Without describing all the experiments in detail, the discussion below presents some illustrative  examples  covering almost all the essential different cases. 

More precisely, the phenomena predicted in rows 1 to 4 of Table \ref{table:CMexs} are illustrated in
Examples \ref{ex-91b-100-split}, \ref{ex-88a-100-nonsplit}, \ref{ex-56b-010-split} and \ref{ex-30a-010-nonsplit}; rows 5 and 6
in Example \ref{ex-77-001-111}; and most cases in rows 7 to 9 in Example \ref{ex-136-100-010}. These examples also illustrate
the unproven predictions for $\lambdahida$ and partial predictions for $\lambdanonhida$ which are evoked in Section \ref{sec-exs-dihedralCM-Hida}.

We begin by considering
 two Asai rank $(1,0,0)$ examples in which the $\sigma_p$-eigenvalues are $(1,-1,(\zeta_3,\zeta_3^{-1}))$, one where the elliptic curve has split multiplicative reduction at $p$ and the other where it has non-split reduction.

 

\begin{example}\label{ex-91b-100-split}  
Let $D_1 := -31$ and $D_2 :=-67$. The field $K_1 = \Q(\sqrt{-31})$ has class number $3$, while
$K_2 = \Q(\sqrt{-67})$ has class number $1$, and $F = \Q(\sqrt{31 \cdot 67})$ has class number $1$
but narrow class number $2$.  Let 
$$\psi :{\rm Cl}_{K/F} \rightarrow
L^\times, \qquad L:= \Q(\zeta_3) $$ be a character of order $3$ of the biquadratic field $K$.

Let
$$\eta := 22  + \left(\frac{1 + \sqrt{31 \cdot 67}}{2}\right) \in O_F, \qquad 
{\rm Norm}(\eta) = -13, \qquad M=13.$$
Choose $p := 7$, which is
inert in $F$.
Note that $S_2(7)= S_2(13) = \{0\}$, and in particular  there are no elliptic curves of conductor $7$ or $13$.

Let $E$ be the elliptic curve  of conductor $91=7\cdot 13$ labelled $91b$ in Cremona's tables, which    is given by the equation
\[ E: y^2 + y = x^3 + x^2 - 7 x + 5.\]
It corresponds to a newform
 $f_E \in S_2(91)$. 
 Letting $H$ denote the Hilbert class field of $K$, the multiplicities with which
the Asai representation $V$ occur within the Mordell-Weil group $E(H)_L$ are
$(1,0,0)$. That is to say, $E$ has rank
$1$ over $\Q$, and this  rank does not increase over $\Q(\sqrt{-67})$ or over any of the
cubic subfields of $H$ of discriminant $-31$.

Furthermore, $E$ has split multiplicative reduction at $7$, so $a_7(E) = +1$. One finds
\[
\lambdatheta = 
 (1 - \zeta_3) \cdot \lambdahida = \lambdanonhida = -4184884843330974 \cdot 7 \pmod{7^{20}}.\] %% above value scaled by 1/3 
Let $P^+ = (3,-5)$ be a generator for $E(\Q)$. Then one checks that
\[  \lambdatheta =  \logtate(P^+)   \pmod{7^{20}}.\]
\end{example} 

\begin{example}\label{ex-88a-100-nonsplit} %% example 47
Set $D_1 := -83$ and $D_2 := -67$, and  take
$$\eta :=1585843 - 41969 \left(\frac{1 + \sqrt{83 \cdot 67}}{2}\right) \in O_F, \qquad {\rm Norm}(\eta) = -8,\qquad M=8.$$
The prime $p := 11$ is inert in $F$, and there is an elliptic curve  $E$ of conductor $88$, 
labelled $88a$ in Cremona's tables,
\[ E: y^2 = x^3 - 4 x + 4.\]
This elliptic curve now has non-split multiplicative reduction at $p$, i.e., $a_p(E) = -1$. 
It has rank one over $\Q$, 
with Mordell-Weil group generated by  $P^+ = (2,-2)$, 
and just as in the
preceding example this rank stays the same  over $\Q(\sqrt{-67})$ or any of the
cubic subfields of $H$ of discriminant $-83$.

A numerical calculation shows that
\[ \lambdatheta =\lambdahida =  0 \pmod{11^{20}}\]
exactly as predicted by the theory, and a further calculation reveals the tantalising numerical identity
\[  \lambdanonhida = -5 \cdot \sqrt{-3} \cdot \logtate(P^+) \pmod{11^{20}}.\]
\end{example}

The  next two examples examine settings where the  Mordell-Weil ranks are $(0,1,0)$,   the  $\sigma_p$-eigenvalues are $(1,-1,(\zeta_3,\zeta_3^{-1}))$,
and $E$ has either  split or non-split multiplicative reduction at $p$.

\begin{example}\label{ex-56b-010-split} %% Example 39
Let $D_1 := -83$ and $D_2 := -11$ and  pick
 $$\eta := 745 + 51 \left(\frac{1 + \sqrt{83 \cdot 11}}{2}\right) \in O_F, \qquad {\rm Norm}(\eta) = -8, \qquad M=8.$$
The prime $p := 7$ is inert in $F$ and also in $K_2$.


 Let $E$ the elliptic
curve of conductor $56=7\cdot 8$ labelled $56b$ in Cremona's tables:
\[  E:y^2 = x^3 - x^2 - 4 .\]
It has rank $0$ over $\Q$, but rank $1$ over $K_2 = \Q(\sqrt{-11})$, and
again rank $0$ over any the cubic subfields of $H$ of discriminant $-83$.
Furthermore $E$ has split reduction at $p$. As predicted,  a calculation shows that
\[  \lambda_{{\theta}} = \lambdahida =  0 \pmod{7^{20}}.\]
The relevant point now to consider is $P^- := (-7,6 \sqrt{-11}) \in E(K_2)$.
We compute
\[ \lambdanonhida = (5563318767325300 c + 5291061116602757) \cdot 7 \pmod{7^{20}},\] 
where $\Q_{p^2} = \Q_p(c)$ with $c^2 + 6 c + 3 = 0$, and  find
\[  \lambdanonhida  = \frac{1}{2} \cdot \sqrt{-3} \cdot \logtate(P^-) \pmod{7^{20}}.\]
\end{example}

\begin{example}\label{ex-30a-010-nonsplit} %% Example 9
Let $D_1 := -31$ and $D_2 := -7$ and choose
 $$\eta := -1888 -275 \left(\frac{1 + \sqrt{31 \cdot 7}}{2}\right) \in O_F, 
 \qquad {\rm Norm}(\eta) = -6, \qquad M=6. $$
  Let $p := 5$, which is inert in $F$ and also in $K_2 = \Q(\sqrt{-7})$. 
  
  We take $E$ the elliptic
curve labelled $30a$ in Cremona's tables,
\[ E:y^2 + x y + y = x^3 + x + 2.\]
This curve has non-split multiplicative reduction at $p$.
Just as in Example \ref{ex-56b-010-split} we have Asai ranks $(0,1,0)$ and the relevant point to consider is
\[ P^- := ((-3 \sqrt{-7} - 9)/8 , (-3 \sqrt{-7} + 31)/16) \in E(K_2).\]
We verify that
\[ \lambdatheta =
(1 + \zeta_3) \cdot \lambdahida = \lambdanonhida =   5248359978986 \cdot 5 \pmod{5^{20}},\]
and that
\[  \lambdatheta = \logtate(P^-) \pmod{5^{20}}.\]
\end{example}

The following experiments consider  two Asai rank $(0,0,1)$ examples, which necessarily have $\sigma_p$-eigenvalues $(1,1,(1,-1))$), when $a_p(E) =1$ and $-1$. These are the most interesting
since  the irreducible component of the Asai representation which occurs with multiplicity one in the Mordell-Weil
group is two-dimensional,  cuts out a cubic abelian extension of $K_1$, and supports points in both the plus and minus eigenspace for $\sigma_p$. 
 

 \begin{example}\label{ex-77-001-111} %% Example 6
Let $D_1 := -23$ and $D_2 := -43$ and 
$$\eta :=  16 - \left(\frac{1 + \sqrt{23 \cdot 43}}{2}\right) \in O_F, \qquad {\rm Norm}(\eta) = -7,\qquad M=7.$$
Let $p := 11$, which is inert in $F$. 

 Although $S_2(M) = \{0\}$, 
there are three elliptic curves of conductor $77 = p M$, which are
labelled $77a,77b,77c$ in the tables of Cremona:
\[
\begin{array}{l}
E_a: y^2 + y = x^3 + 2 x \\
E_b: y^2 + y = x^3 + x^2 - 49 x + 600 \\
E_c: y^2 + x y = x^3 + x^2 + 4 x + 11.
\end{array}
\]
These curves have non-split, non-split and split multiplicative reduction at $p=11$, respectively.

Recall that $H$ is the Hilbert class field of the quartic field $K = F(\sqrt{-23})$, and that the 
subfields cut out by the  Asai representation $V$ are $K_2 = \Q(\sqrt{-43})$ (of class number one), and the Galois closure of 
any of the cubic subfields of $H$ of conductor $-23$. (These are also subfields of the Hilbert
class field $H_1$ of the class number three field $K_1 = \Q(\sqrt{-23})$.) The cubic subfield is isomorphic to  $\Q(u)$ where $u^3 - 3 u^2 - 23 = 0$.

The Asai ranks for the elliptic curve  $E_a$ are $(1,1,1)$, i.e., $E_a$ has rank
$1$ over $\Q$, and its rank jumps to $2$ over both $K_2$ and the cubic subfield $\Q(u)$ of conductor $-23$.
As expected, we observe that 
\[ \lambda_{{\theta}} =
\lambdahida = \lambdanonhida = 0 \pmod{11^{20}}, \]
so that all three invariants 
 vanish up to the computed precision. 
 
 Turning next to 
 $E_b$, which has non-split reduction at $11$,  the Asai
ranks are equal to $(0,0,1)$. Let
\[ Q_W = ((-u^2 + 6 u + 10)/3 , (7 u^2 + 35 u - 71)/9) \in E_b(\Q(u)).\]
The  global 
point $P_W^- \in E(\Q_{p^2})$ in the {\it negative} eigenspace for $\sigma_p$, is obtained as follows. Let $u_1$ and $u_2$ be the two roots of $x^3 - 3 x^2 - 23 = 0$
in $\Q_{p^2}$ which are not in $\Q_p$ and are conjugate to each other. Replacing $u$ in the point $Q_W$ above by $u_1$ and $u_2$
yields two points $Q_1$ and $Q_2$ in $E(\Q_{p^2})$ which  are conjugate under $\sigma_p$.
Define 
$$P_W^- = Q_1 - Q_2.$$
Numerically, one finds
\[ 2 \cdot \lambda_{{\theta}} = \lambdahida = 14646999780697863202 \cdot 11 \pmod{11^{20}},\]
and  
\[  2 \cdot \lambdatheta = \lambdahida = -\frac{1}{5} \cdot \logtate(P_W^-) \pmod{11^{20}}.\]

Finally, the elliptic curve $E_c$ has split reduction at $11$ and  Asai ranks $(0,0,1)$. The global point on $E_c$ over the cubic subfield of the Hilbert  class field  of $K_1$ is equal to 
\[ Q_W = ((2 u^2 - 5 u - 7)/9 , (-u^2 - u + 9)/3) \in E_c(\Q(u)).\]
The point $P_W^+ \in E(\Q_{p})$ in the {\it positive} eigenspace for  $\sigma_p$ is obtained from $Q_W$ by mapping $u$ to the unique root  of $x^3 - 3 x^2 - 23$ that lies in $\Q_p$.

Numerically we verify that
$$\lambda_{{\theta}} = 0 \pmod{11^{20}},$$ 
as predicted 
since $a_p(E_c) = +1$ occurs with multiplicity
$3$ as an eigenvalue for $\sigma_p$ on $V$. 
On the other hand,
\[ \lambdahida   = (-10945417843758550651 a - 7729417228947654133) \cdot 11 \pmod{11^{20}},\]
 where $a^2 + 7 a + 2 = 0$, and
\[  \lambdahida = \frac{1}{2} \cdot \sqrt{-3} \cdot \logtate(P_W^+) \pmod{11^{20}},\]
as expected since $a_p(E_c)$ has multiplicity $1$ in the component $W$ of the Asai representation
which   occurs with rank $1$ in the Mordell-Weil group. (For $E_a$ and $E_b$ the value
$\lambdanonhida$ is non-zero, but we have no interpretation of it to offer.)
\end{example}

%% Finally we consider an Asai rank $(0,1,0)$ example, but where the Frobenius eigenvalues are $(1,-1,(1,1))$.

%% \begin{example}\label{ex-34a-010-nonsplit}
%% We let $D_1 := -59$ and $D_2 := -11$ and
%% \[ \eta :=  2129939  -160899\left(\frac{1 + \sqrt{59 \cdot 11}}{2}\right) \in O_F,\]
%% an element of norm $-2$ in $F = \Q(\sqrt{59 \cdot 11})$. Let $p := 17$ and
%% $E$ be the elliptic curve labelled $34a$:
%% \[ E: y^2 + x y = x^3 - 3 x + 1.\]
%% We have non-split reduction at $p$ and Asai ranks $(0,1,0)$. Letting $P^- := (3/4 , (2 \sqrt{-11} - 3)/8) \in E(K_2)$
%% be a generator for the Mordell-Weil group (modulo torsion) we find
%% \[ \lambdatheta = 2 \cdot \lambdahida = 2 \cdot \lambdanonhida = 8646140326 \cdot 17 = \logtate(P^-) \bmod{17^{10}}.\]
%% \end{example}

Finally we consider Asai rank $(1,0,0)$ and $(0,1,0)$ examples, but where the $\sigma_p$-eigenvalues are $(1,-1,(1,1))$.
The smallest prime occurring in any example in this setting is $p = 17$.

\begin{example}\label{ex-136-100-010}
Let $D_1 := -59$ and $D_2 := -11$ and
$$\eta :=  35786444535052398935  -2703365279120236257\left(\frac{1 + \sqrt{59 \cdot 11}}{2}\right) \in O_F, $$
so that $ M=  - {\rm Norm}(\eta) = 8$.

 Letting $p :=17$ we find the $\sigma_p$-eigenvalues on the Asai components are $(1,-1,(1,1))$.

First consider the curve of conductor $Mp$ labelled $136a$:
\[ E_a: y^2 = x^3 + x^2 - 4 x.\]
It has ranks $(1,0,0)$ and split reduction at $p$.
 But since $+1$ also occurs  as an eigenvalue
in the dimension $2$ component of the Asai representation, Theorem \ref{conj-CM} 
predicts that $\lambdatheta = 0$ and
we indeed observe this numerically (to  precision $17^{10}$). However, since $+1$ occurs with
multiplicity one in the component of the Asai representation which itself has rank one in the Mordell-Weil group,
we expect $\lambdahida$ to be more interesting, and indeed find that
\[ \lambdahida = (-8218050809 a + 52843470025) \cdot 17 \pmod{17^{10}} \]
where $a^2  + 16 a + 3 = 0$, and that
\[ \lambdahida = \frac{2 \zeta_3 + 1}{2} \cdot \logtate(P^+) \pmod{17^{10}}\]
where $P^+ = (2,-2)$ generates $E_a(\Q)$ modulo torsion. On the other hand we also find that $\lambdanonhida = 0 \pmod{17^{10}}$.

\medskip
Consider next the curve labelled $136b$:
\[ E_b: y^2 = x^3 - x^2 - 8 x - 4.\]
This has ranks $(0,1,0)$ and non-split reduction at $p$. We find as expected that
\[ \lambdatheta :=-55227507250 \cdot 17 = \lambdanonhida = 2 \cdot \lambdahida =  \frac{1}{2} \cdot \logtate(P^-) \pmod{17^{10}},\]
where $P^- = (-430/891 , -18512\sqrt{-11}/88209) \in E_b(K_2)$ is a generator modulo torsion. 
\end{example}

\subsubsection{Further examples for CM biquadratic extensions}

We computed some further examples in settings very close to that outlined in Section \ref{sec-exs-dihedralCM-exs}. Namely, when $K_1$ had class number $5$ and
$K_2$ class number $1$, and when $K_1$ had class number $5$ and $K_2$ class number $3$. We spare the reader any details, beyond saying that
they were in agreement with our theorem for the theta derivation coefficient $\lambdatheta$ and in line with our expectation
(from Section \ref{sec-exs-dihedralCM-Hida}) for the Hida derivation coefficient $\lambdahida$.

 



\subsection{Biquadratic extensions: the RM setting}\label{sec-exs-dihedralRM}

Let $D_1 = \ell \equiv 1 \bmod{4}$ and $D_2 = q \times r $ where $\ell,q,r$ are distinct primes with
$q,r \equiv 3 \bmod{4}$. We assume $\Q(\sqrt{D_1})$ has narrow class number $1$, and $\Q(\sqrt{D_2})$ has
class number $1$ but narrow class number $2$.

Define 
\[ F = \Q(\sqrt{D_1 D_2}) \mbox{ and } K = \Q(\sqrt{D_1}, \sqrt{D_2}).\]
Then $K$ is an unramified totally real extension of $F$ which is Galois over $\Q$.
The field $F$ has
class number $2$ and narrow class number $4$.

There is a mixed signature Hecke character $\psi$ on $K$ of order $2$ with trivial finite conductor $O_F$.
By induction one obtains as before a (totally odd) $2$-dimensional Galois representation $V_0$ of $F$.
The four dimensional Asai induction of $V_0$ decomposes as
\[ {\rm Asai}(V_0) = {\bf 1} \oplus \varepsilon_\ell \oplus \varepsilon_q \oplus \varepsilon_r \]
where  $\varepsilon_\ell, \varepsilon_q, \varepsilon_r$ are quadratic characters attached to the
fields $\Q(\sqrt{\ell}), \Q(\sqrt{-q})$ and $\Q(\sqrt{-r})$ respectively.

After choosing an ideal $(\eta)\subset \cO_F$ which is trivial in the class group but non-trivial in the narrow class group, i.e.
an element $\eta$ of negative norm, one obtains the Fourier expansion of a Hilbert modular form $\theta_\psi$ on $F$ of weight $(1,1)$.
Note that here as $K$ is the only unramified extension of $F$, there cannot exist a CM extension
of $F$ and Hecke character on that CM extension which defines a Hilbert Modular Form of trivial
level $O_F$. So in particular, the theta series $\theta_\psi$ has RM but no CM.

Let $p$ be a prime which is  inert in $F$. We observe that always then $p O_K = P P'$
for distinct prime ideals in $O_K$. Let $E$ be an elliptic curve of conductor $pM$, where $M:= |{\rm Norm}(\eta)|$.
Just as in the CM dihedral examples of Section
\ref{sec:bi-CM},  the possible patterns of
ranks of the Asai components in the Mordell-Weil group and corresponding eigenvalues of
$\sigma_p$ on these components are correlated  with the pair $(\psi(P),a_p(E))$. 
Without going into too much detail,  observe that  Conjecture 
\ref{main-conj} only predicts the non-vanishing
of $\lambdatheta$ in the $25 \%$ of times in which $(\psi(P),a_p(E)) = (+1,-1)$. We give below
a selection of examples in that setting (our coefficients vanished experimentally in the other settings, as expected).

Note that these examples give evidence for our conjecture beyond what can be proved in our main theorem (Theorem \ref{conj-CM}).
Our theorem only predicts ``$\lambdatheta = \logtate(P)$'' for a Stark-Heegner point $P$, and these are
only known in this setting to be local points. Our examples show $P$ is a global point.

%% The numbers for these examples are in logfiles2023/2024_Sept20_RM_no_CM_Juneexrs_m20.m
%% The analysis, with the points, in DLR-rk1-14/13_57_eta3_p7_RM_200924.m

%% NOTE: I expect the values are off by sign for non-split ones because I probably didn't account for the hitting by U_p

\begin{example}
Let 
$$D_1 := 13, \quad D_2 := 57 = 3 \times 19, 
\qquad
 \eta = -14 - \left(\frac{1 + \sqrt{13 \cdot 57}}{2}\right) \in O_F, \quad M=3,$$
 and choose
$p := 7$, which is inert in
$F$. 
We take 
$$E: y^2 + x y = x^3 - 4 x - 1$$
 to be the elliptic curve labelled $21a$ in the tables of Cremona.
This curve has non-split multiplicative reduction at $p$.
 Let 
 $$P^- = ((-\sqrt{13} + 3)/2,\sqrt{13} - 3) \in E(\Q(\sqrt{13})).$$
Then
\[ \lambdatheta = 903550508804407 \times 7 = -4 \cdot \logtate(P^-) \pmod{7^{20}}.\]
\end{example}

\begin{example}
Let 
$$D_1 := 13, \quad D_2 := 93 = 3 \times 31,
\qquad \eta = 8983 + 532 \left(\frac{1 + \sqrt{13 \cdot 93}}{2}\right) \in O_F ,$$
so that $M=3$. 
Choose
 $p := 7$. We again take $E$ to be the elliptic curve $21a$ and $P^-$ as in the preceding example.
Then we have
\[ \lambdatheta = -903550508804407 \times 7 = 4 \cdot \logtate(P^-) \pmod{7^{20}}.\]
\end{example}

\begin{example}
Let 
$$D_1 := 5, \quad D_2 := 93 = 3 \times 31,$$
and choose
$$ \eta = 10 + \left(\frac{1 + \sqrt{5 \cdot 93}}{2}\right) \in O_F, \qquad M= 6, \quad 
p := 7.$$
Let 
$$E: y^2 + x y + y = x^3 + x^2 - 4 x + 5$$ 
 be the elliptic curve $42a$, which has non-split multiplicative reduction at $p$.
After defining 
$$P^- = (- \sqrt{5}, 2 \sqrt{5} - 1) \in E(\Q(\sqrt{5})),$$
we observe that
\[ \lambdatheta = 3617170189375121 \times 7 = \lambdatheta = 2 \cdot \logtate(P^-) \pmod{7^{20}}.\]
\end{example}

\begin{example}
Let 
$$D_1 := 193, \quad D_2 = 33 = 3 \times 11,$$
and choose
$$ \eta = -2707 + 67\left(\frac{1 + \sqrt{193 \cdot 33}}{2}\right) \in O_F, \qquad M=8, \qquad p:=7.$$
We take 
$$E: y^2 = x^3 + x + 2$$
 to be the elliptic curve $56a$,
which has non-split multiplicative reduction at $p$, and 
$$P^- = (-26, 2 \sqrt{-11})  \in E(\Q(\sqrt{-11})).$$
Then
\[ \lambdatheta = 3274774510128699 \times 7 = -2 \cdot \logtate(P^-) \pmod{7^{20}}.\]
\end{example}

\begin{example}
Let 
$$D_1 := 181, \qquad D_2 = 21 = 3 \times 7,$$
and choose
$$ \eta := -5 -1441\left(\frac{1 + \sqrt{181 \cdot 21}}{2}\right) \in O_F, \qquad M=5,  \qquad p := 11. $$
 We take 
 $$E: y^2 + x y = x^3 - x^2 - 4 x + 3$$ to be the elliptic curve labelled $55a$ in the tables of Cremona.
This curve has non-split multiplicative reduction at $p$. Define 
$$P^- = (1, (-\sqrt{-3} - 1)/2) \in E(\Q(\sqrt{-3})).$$
Then
\[ \lambdatheta = 834444252005378646 \times 11 = 2 \cdot \logtate(P^-) \pmod{11^{20}}.\]
\end{example}

Observe that in the RM case the Hida coefficients $\lambdahida$ from Section \ref {sec-exs-dihedralCM-Hida} may still be defined, but we do not know how to efficiently compute them.



\subsection{A cyclic quartic extension}
\label{subsec:cyclic-quartic}
We now give an example which is beyond the reach of our current theorem and the machinery of
Heegner and Stark-Heegner points (though still within the purview of CM extensions).



\begin{example}\label{ex-cyclicquartic}
Let $F := \Q(\sqrt{17})$ and 
\[ \alpha := -8 - \frac{1 + \sqrt{17}}{2}.\]
Then $K := F(\sqrt{\alpha})$ is a CM extension of $F$ which is a cyclic quartic extension
of the rational field. We have $17 \cdot O_K = P^4$ for a prime ideal $P$. The group of
Hecke characters of $K$ of conductor $P$ is $\Z/2\Z\times \Z/8\Z$ and we take
$\psi$ to be an element of order $4$ in this group.

The diagonal restriction of the theta series $\theta_\psi$ attached to $\psi$ lies in $S_2(4 \times 17)$, a space of
dimension $7$. (Here $F$ has narrow class number $1$ and the Fourier expansion for $\theta_\psi$
is found using the trivial ideal $O_F$.)
Note that the diagonal restriction of $\theta_\psi$ is killed by the Hecke operators
$U_2(U_2 - I)$ and $U_{17} + I$.

Let $p:=5$, which is inert in $F$. Using our computational methods, it is possible then to
consider all elliptic curves of conductor divisibly by $5$ and dividing $5 \times 68$. In the Magma
computer algebra package these
have Cremona labels
\[ 20a, 85a, 170a, 170b, 170c, 170d, 170e, 340a .\]
(Beware however that this labelling of the curves in Magma of conductor $170$ is not the
same as that in \cite{LMFDB}.) Letting $V_0$ be the induction of $\psi$ to $G_F$ and $V$ its Asai induction to $\Q$, the representation $V$ is self-dual and decomposes as a direct sum
\[ {\rm Asai}(V_0) = \chi_4 \oplus \chi_{68} \oplus \eta_{17} \oplus \overline{\eta}_{17}.\]
Here $\chi_4$ and $\chi_{68}$ are quadratic characters of conductors $4$ and $68$, and
$\eta_{17}$ a quartic character of conductor $17$. Using Magma we compute that the order of
vanishing of $L(E,{\rm Asai} (V_0))(s)$ at $s = 1$ is (very likely to be) one when $E$
is the curve $20a$, $170c$ or $170d$, and even otherwise.

Because the diagonal restriction of the Hilbert modular theta series $\theta_\psi$ is non-zero, we need to modify our theta derivative
using a Hecke operator which kills the $p$-depletion of this non-zero form. In practice we must use $U_2(U_2 - I)$ or $U_{17} + I$. 
(The modification is done on Fourier expansions themselves, and is very time-consuming, thus limiting
the choice to the simplest possibilities.)

Let $f_{20a}$, $f_{170c}$ and $f_{170d}$ denote the weight two eigenforms attached to the three remaining curves. 
We observe $U_2(U_2 - I)$ kills $f_{20a}$ and $f_{170c}$, and $U_{17} + I$ kills $f_{170c}$ and $f_{170d}$.
In particular, we cannot extract any non-zero values for the curve labelled  $170c$ using these modifications. We shall
thus focus on the curves with labels $20a$ and $170d$.

We consider first 
\[ E_{20a} : y^2 = x^3 + x^2 + 4 x + 4,\]
which has non-split multiplicative reduction at
$p = 5$. The rank one component of Mordell-Weil
appearing in the Asai representation is generated by $$P^-:= (-53/4 , 91 \sqrt{-17} / 8).$$ 
The frobenius element 
 $\sigma_p$ acts by $-1$ on this point,
and we find that
%% \footnote{AL to HD/VR: We take the test-vector here to be the unique $\tilde{f} \in \langle f_{20a}(q), f_{20a}(q^{17}) \rangle \subset S_2(340)$ with
%% $(U_{17} + I)(\tilde{f}) = f_{20a}$. For $170d$ below, we take $\tilde{f} = f_{170d}$, which is a full eigenform in $S_2(340)$.
%% AL: I will add this to the body of the text later, once I can refer back to something for the notation $\tilde{f}$.}
\[ \lambdatheta  = 8 \cdot \logtate(P^-) \pmod{5^8}.\]
Working with the Hida rather than theta derivation, we can get slightly higher precision and find
\[  \lambdahida  = 337558 \cdot 5 = 8 \cdot \logtate(P^-) \pmod{5^{10}}.\]
Next consider 
\[ E_{170d}: y^2 + x y + y = x^3 - 3 x + 6,\]
 which has split multiplicative reduction at $p$. Here the rank one component of
Mordell-Weil appearing in $V$ is generated by
 $$P^+ := (-5, -10 \sqrt{-1} + 2), $$
 and 
 $\sigma_p$ acts by $+1$ on this point. We find
 \[ \lambdatheta = 8 \cdot \logtate(P^+) \pmod{5^9}.  \]
Working with the Hida rather than theta derivation, we can again get higher precision and find
 \[ \lambdahida = 818912872  \cdot  5^2 = 8 \cdot \logtate(P^+) \pmod{5^{16}}.  \]
Observe our precisions for $\lambdatheta$ are much lower than in previous examples. This is because we had to
further act upon the Fourier expansion of the theta derivative by small powers of $U_p$, to ``improve overconvergence''.
% We do not understand rates of overconvergence for these derivatives well enough to know why, but it is
%severe limitation of our computational reach. 
\end{example}

\begin{remark} The modular form $f_{20a}$ is not a full eigenform in $S_2(340)$, and there
is not a canonical choice of ``test vector'' in the old space $\langle f_{20a}(q), f_{20a}(q^{17}) \rangle \subset S_2(340)$ it generates to
take the coefficient along. Given our modification by $U_{17} + I$ it was natural and convenient to choose the unique $\tilde{f}$ in this space with $(U_{17} + I)(\tilde{f}) = f_{20a}$. 
%% For $170d$, we just take $\tilde{f} = f_{170d}$, which is a full eigenform in $S_2(340)$.
\end{remark}
%% 16/12/25: for 20a in fact I project on to f_20a, but this is after modifying the form to be projected by U_17 + I.
%% I think for 170d we I used the fact that f_170a is a U_2 eigenform with eigenvalue -1, and modification by
%% U_2(U_2 - I) just becomes scaling by 2, and I accounted for this.

\subsection{A non-Galois quartic extension}
\label{sec:non-galois}
Our final example is beyond both the reach of Heegner and Stark-Heegner points, and even outside that of CM extensions.

\begin{example}
Let $F = \Q(\sqrt{301})$ and 
\[ \alpha:= \frac{-\sqrt{301}- 17}{2},\]
which has norm $-3$. We have $3 \cdot O_F = P P^\prime$
where $P = (\alpha)$ and $P^\prime = (\alpha')$ with
$\alpha' := (\sqrt{301} - 17)/2$.
Define $K := F(\sqrt{\alpha})$ and
\[ \beta := \frac{5 \sqrt{301} - 87}{4} \sqrt{\alpha} + \frac{\sqrt{301} - 18}{2} \in K,\quad {\rm Norm}_{K/F}(\beta) = \alpha'.
\]

The field $K$ is a non-Galois quartic extension of signature $(2,1)$, with
Galois closure a degree $8$ extension of $\Q$.
There is a Hecke character $\psi$ of order $2$  of $K$ factoring through the extension
$K(\sqrt{\beta})$ such that the diagonal restriction of the associated series
$\theta_\psi$ (constructed using the trivial ideal $O_F$) lies in $S_2(3)$; in particular, it has  {\it trivial} character. (Finding
non-Galois quartic extensions and Hecke characters where the diagonal restriction
has quadratic character is much easier. The point of this example is that the
character is trivial.) Very conveniently $S_2(3)$ is zero, and so this diagonal restriction must vanish. Thus we
do not need to modify our theta derivative by any Hecke operators.

Letting $V_0$ be the induction of $\psi$ to $G_F$, the Asai representation itself is self-dual and decomposes as a direct sum
\[ {\rm Asai}(V_0)  = 1 \oplus \chi_3 \oplus W_{903}\]
where $\chi_3$ is a quadratic character of conductor $3$ and $W_{903}$ an irreducible
$2$-dimensional representation of conductor $903$.

We consider small primes $p$ which are inert in $F$ and for which we have
curves in levels $p$ or $3 p$. First, we take $p := 13$ and consider
the curve $E_{39a}$ labelled $39a$ by Cremona. The ranks
of the components of the Asai representation in the Mordell-Weil group are   $(0,1,2)$.
Since we are in a rank $3$ setting, Conjecture 
\ref{main-conj} predicts the component along the attached weight
two form $f_{39a}$ of the theta derivative should be zero. And we observe this numerically (to precision $13^5$).

Now let $p:= 17$. Here we may consider two curves, labelled
$17a$ and $51a$. The ranks in the Asai representation of the Mordell-Weil group
are respectively $(0,0,1)$ and $(0,1,0)$. The $\sigma_p$-eigenvalues
on the three components of ${\rm Asai}(V_0)$ are $+1,-1$ and $\pm i$. The curve labelled
$17a$ has split multiplicative reduction at $p$, but in any case $+1$ (or $-1$)
does not occur as an $\sigma_p$-eigenvalue on $W_{903}$. So our conjecture
predicts the component of the theta derivative along $f_{17a}$ should be
zero, as we observe numerically (to precision $17^5$).

More interestingly, the other curve 
\[ E_{51a} : y^2 + y = x^3 + x^2 + x - 1\] has non-split reduction
at $p$, and here $-1$ does occur as an eigenvalue on the rank $1$ component.
The relevant generator is $$P^- = ((\sqrt{-3} - 1)/2 , (\sqrt{-3} - 1)/2) \in E_{51a}(\Q(\sqrt{-3})),$$ and we find
\[ \lambdatheta = 734096 \cdot 17 = 4 \cdot \logtate(P^-) \pmod{17^6}.\]
\end{example}

\subsection{Beyond Hecke characters}
Let $g$ be an exotic modular form of weight $1$ and quadratic character
$\chi$. Given a real quadratic field $F$, one may restrict the Galois representation
$\rho_g$ of $\Q$ to $F$, and then take the Asai induction of that to get a four dimensional representation
${\rm Asai}(\rho_g |_{G_F})$. Since $\chi$ is quadratic we find this Asai representation is self-dual. This is the
most appealing setting in which to do further experimental work, and some partial progress has been made in that direction.

The smallest examples to consider are the exotic forms in levels $283$ and $331$, which have projective image $S_4$. (See for example the webpage accompanying \cite{BL}, or
\cite{LMFDB}.) For each of these, using 
code for computing Hilbert class fields in Magma we were able to find the degree $48$ fields through which each of
the {\it linear} representations for these forms factors. Then given $F = \Q(\sqrt{D_F})$ with $D_F = 1 \bmod{4}$ (and
$D_F$ inert in $\Q(\sqrt{-283})$ or $\Q(\sqrt{-331})$, respectively), we could find both the diagonal restriction of
the theta family and   of its derivative. The (so far) insurmountable  difficulty we then encounter
is that the diagonal restriction of the theta family lies in $S_2(283)$ (or $S_2(331)$ respectively), which is a space
of dimension $23$ (respectively $27$). And it seems difficult in practice to modify the family by a Hecke operator
on Fourier expansions to kill a $p$-depletion of this form, a necessary step to move from the largely mysterious world
of ``$p$-adic mock modular forms'' to that of (nearly) overconvergent ones. Note that forms in latter infinite dimensional
spaces are far more amenable to computation, using the methods of \cite{AL1,AL2}.
It is evident that a better theoretical and
computational understanding of $p$-adic mock modular forms is required in order to push the experimental work
further.


\medskip\medskip\noindent
{\small
 \thanks{{\bf Acknowledgements.} The first author was supported by an NSERC Discovery grant, and the
  third author was supported by Grant MTM2015-63829-P.  
 The third author is supported by Icrea through an Icrea Academia Grant, Proyecto de generaci\'on de conocimiento PID2022-137605NB- I00 and by the Spanish State Research Agency through the
Severo Ochoa and Mar\'ia de Maeztu Program for Centers and Units of Excellence in R\& D (CEX2020-
001084-M). The experimental work was carried out by the second author using the Magma Computational Algebra System.

The authors are most thankful to Ananyo Kazi for his interest and comments, which prompted us to improve the exposition. 
}}





%% Unused references

%% \bibitem[CV]{CV} 
%% S. Cho, V. Vatsal, {\em Deformations of induced Galois representations}, J. reine angew. Math. {\bf 556}, (2003), 79-97.


%% \bibitem[BSV2]{BSV-Ast2}
%% M. Bertolini, M. A. Seveso, R. Venerucci, second paper in Asterisque


%% \bibitem[DLR2]{DLR2}
%% H.~Darmon, A. Lauder, V. Rotger,  
%% {\em Gross-Stark units and  $p$-adic iterated integrals attached to modular forms of weight one}, Annales
%% Math\'ematiques du Qu\'ebec {\bf 40} (2016), Issue 2, 325-354.


%% \bibitem[DLR3]{DLR3}
%% H. Darmon, A. Lauder,  and V. Rotger, 
%% {\em
%% Overconvergent generalised eigenforms of weight one and class fields of real quadratic fields}. 
%% Advances in Mathematics {\bf 283} (2015), 130-142.

%% \bibitem[DLR4]{DLR4}
%% H.~Darmon, A. Lauder, V. Rotger,  
%% {\em First order  $p$-adic deformations of  weight one newforms},
%% in {\em Heidelberg conference on $L$-functions and automorphic forms}, Contrib. Math. Comp. Sci. {\bf 10}, Springer, (2017), 39--80.


%% \bibitem[EP]{EP}
%% H. Edgar, B. Peterson, {\em Some Contributions to the Theory of Cyclic Quartic Extensions of the Rationals}, J. Number Theory {\bf 12}, 77--83 (1980).

%% \bibitem[Gr]{Gr} G. Graziani, {\em Modular sheaves of de Rham classes on Hilbert formal modular schemes for unramified primes}, available at {\em arXiv:2007.15997}.

%% \bibitem[HaKu]{HaKu}
%% M. Harris, S. Kudla, Annals


%% \bibitem[Hi]{Hi}
%% H. Hida, {\em On $p$-adic Hecke algebras for $\GL_2$ over totally real fields}, Ann. Math. {\bf 128} (1988), 295--384.

%% \bibitem[Hi]{Hi}
%% H. Hida, {\em On $p$-adic Hecke algebras for $\GL_2$ over totally real fields}, Ann. Math. {\bf 128} (1988), 295--384.

%%  \bibitem[Liu]{Liu} Y. Liu, {\em Hirzebruch-Zagier cycles and twisted triple product Selmer groups}, Invent. Math. {\bf 205} (2016), no. 3, 693--780.

%% \bibitem[CM]{ref-Heegner}
%% Some paper on Heegner points

%% \bibitem[AIP]{AIP} 
%% F. Andreatta, A. Iovita, V. Pilloni, {\em On overconvergent Hilbert modular cusp forms}, in {\em Arithm\'etique p-adique des formes de Hilbert} (2016), 163--193, Soci\'et\'e Math\'ematique de France.

%% \bibitem[Ay]{Ay} J. Aycock, {\em Families of differential operators for overconvergent Hilbert modular forms}, available at {\em arxiv.org/pdf/2006.02543}.


%%  \bibitem[BDi1]{BeDi1} 
%% J. Bella\"iche, M. Dimitrov, {\em On the eigencurve at classical weight one points},
%% Duke Math. J.  {\bf 165 } (2016), no. 2, 245--266.

%% \bibitem[BDi2]{BeDi2} A. Betina, M. Dimitrov, {\em Geometry of the eigencurve at CM points and trivial zeros of Katz $p$-adic $L$-functions}, Adv. Math. {\bf 384} (2021).
 
%% \bibitem[Bet]{Betina} A.~Betina,  {\em Ramification of the eigencurve at classical RM points}, Canad. J. Math. {\bf 72} (2020), no. 1, 57--88.

%% \bibitem[Pr]{Pr}
%% D. Prasad, {\em Invariant forms for representations of $\GL_2$ over a local field}, Amer. J. Math. {\bf 114} (1992), no. 6, 1317--1363.

%% \bibitem[PrSP]{PrSP} D. Prasad, R. Schulze-Pillot, {\em Generalised form of a conjecture of Jacquet and a local consequence}, J.l Reine angew.\,Math., {\bf 616} (2008), 219--236.

%% \bibitem[Se]{Se}
 %% J.-P.\,Serre, {\em Local fields}, Springer-Verlag, 1979.
 
%%  \bibitem[Si]{Si} J. Silliman, {\em Group ring valued Hilbert modular forms}, Res. number theory {\bf 10: 38} (2024).

%% \bibitem[Ta]{Ta} R. Taylor, {\em On Galois representations associated to Hilbert modular forms}, Inventiones Math. {\bf 98} (1989), 265--280.


\end{document}

